

Julián Martínez-Simancas Secretary of the Board of Directors

Bilbao, 17 June 2016

To the National Securities Market Commission

Subject: Shareholders' agreement relating to Gamesa Corporación Tecnológica, S.A.

Dear Sirs,

Pursuant to article 228 of the restated text of the Securities Market Law approved by Royal Legislative Decree 4/2015, of 23 October (texto refundido de la Ley del Mercado de Valores aprobado por el Real Decreto Legislativo 4/2015, de 23 de octubre) and related provisions, and pursuant to article 531 of the restated text of the Spanish Companies Law, approved by Royal Legislative Decree 1/2010, of 2 July (texto refundido de la Ley de Sociedades de Capital, aprobado por el Real Decreto Legislativo 1/2010, de 2 de julio) (the "Spanish Companies Act") we hereby inform you that:

- On the date hereof, Iberdrola, S.A. and Iberdrola Participaciones, S.A. Sociedad Unipersonal¹, as shareholders (indirect and direct, respectively) of Gamesa Corporación Tecnológica, S.A. ("Gamesa"), on the one side, and Siemens AG, on the other side, have entered into an agreement relating to (i) Gamesa in the context of a merger proceeding between the wind power businesses of Gamesa and of Siemens AG (the "Merger") and (ii) to their relationship as future shareholders of Gamesa after the Merger (the "Shareholders' Agreement").
- The Shareholders' Agreement qualifies as an agreement among shareholders (*pacto parasocial*) in accordance with the terms of article 530 of the Spanish Companies Law as a result of certain covenants provided for therein. Even though the effectiveness of certain of said covenants is conditional upon the completion of the Merger, a full copy of the Shareholders' Agreement (drafted in the English language), together with a translation into Spanish for informational purposes only, is attached hereto as **Annex**.
- Likewise, please be advised that the existence and content of the Shareholders' Agreement are subject to communication to Gamesa and deposit with the Commercial Registry of Biscay.

This information is provided to you for the appropriate purposes.

Yours faithfully,

Secretary of the Board of Directors

Wholly-owned subsidiary of Iberdrola, S.A.



_



IMPORTANT INFORMATION

This communication does not constitute an offer to purchase, sell or exchange or the solicitation of an offer to purchase, sell or exchange any securities. The shares of Iberdrola, S.A. may not be offered or sold in the United States of America except pursuant to an effective registration statement under the Securities Act or pursuant to a valid exemption from registration.

This communication contains forward-looking information and statements about Iberdrola, S.A., including financial projections and estimates and their underlying assumptions, statements regarding plans, objectives and expectations with respect to future operations, capital expenditures, synergies, products and services, and statements regarding future performance. Forward-looking statements are statements that are not historical facts and are generally identified by the words "expects", "anticipates", "believes", "intends", "estimates" and similar expressions.

Although Iberdrola, S.A. believes that the expectations reflected in such forward-looking statements are reasonable, investors and holders of Iberdrola, S.A. shares are cautioned that forward-looking information and statements are subject to various risks and uncertainties, many of which are difficult to predict and generally beyond the control of Iberdrola, S.A., that could cause actual results and developments to differ materially from those expressed in, or implied or projected by, the forward-looking information and statements. These risks and uncertainties include those discussed or identified in the public documents sent by Iberdrola, S.A. to the Comisión Nacional del Mercado de Valores.

Forward-looking statements are not guarantees of future performance. They have not been reviewed by the auditors of Iberdrola, S.A. You are cautioned not to place undue reliance on the forward-looking statements, which speak only as of the date they were made. All oral or written forward-looking statements hereby made or otherwise attributable to Iberdrola, S.A. or any of its members, directors, officers, employees or any persons acting on its behalf are expressly qualified on its entirety by the cautionary statement above. All the forwardlooking statements included herein are based on information available to Iberdrola, S.A. on the date hereof. Except as required by applicable law, Iberdrola, S.A. does not undertake any obligation to publicly update or revise any forward-looking statements, whether as a result of new information, future events or otherwise.







Annex

Copy of the Shareholders' Agreement (in English) and Spanish translation of the Shareholders' Agreement

NOTICE: This document is a translation of a duly approved Spanish-language document, and is provided for informational purposes only. In the event of any discrepancy between the text of this translation and the text of the original Spanish-language document which this translation is intended to reflect, the text of the original Spanish-language document shall prevail.



SIEMENS AKTIENGESELLSCHAFT IBERDROLA PARTICIPACIONES, S.A.U. and IBERDROLA, S.A.

AGREEMENT
IN CONNECTION WITH THE MERGER
BETWEEN GAMESA AND SIEMENS WIND POWER PARENT



Table of Contents

Conte	ents Page	
PART	IES1	
WHEREAS1		
l.	GENERAL2	
1	Definitions2	
2	Purpose of this Agreement2	
11.	TRANSACTION-RELATED UNDERTAKINGS	
3	Transaction structure and process3	
4	interim period4	
5	Cooperation4	
6	Standstill and lock-up5	
7	Exclusivity5	
III.	RELATIONSHIP BETWEEN SIEMENS AND IBERDROLA AS SHAREHOLDERS OF GAMESA 6	
8	Corporate governance6	
9	Headquarters and operations10	
10	Minority shareholders protection	
11	Squeeze-out; delisting; package sale of shares11	
12	Breach	
IV.	MISCELLANEOUS14	
13	Representations	
14	Term ,	
15	Termination	
16	Announcements 15	
17	Confidentiality	
18	SIEMENS Wind Power Parent Shareholders17	
19	Shareholders Agreement Steering Committee18	
20	Other provisions	
Sche	dule 1.1 Definitions	



PARTIES

(1) SIEMENS AKTIENGESELLSCHAFT, a stock corporation incorporated in Germany, registered with the Commercial Register (*Handelsregister*) at the local court (*Amtsgericht*) of Munich (Germany) under HRB 6684 and the local court (*Amtsgericht*) of Berlin-Charlottenburg (Germany) under HRB 12300, with registered office at Wittelsbacherplatz, 2, 80333 Munich, Germany ("SIEMENS").

SIEMENS is represented herein by Ms Dagmar Mundani and Dr Christian Zentner, acting jointly in their capacity as authorised signatories (*Prokuristen*).

(2) IBERDROLA PARTICIPACIONES, S.A.U., a public limited company (sociedad anónima) incorporated in Spain, registered with the Commercial Registry of Madrid under volume 33515, sheet 45, page number M-603299, whose registered office is at calle Tomás Redondo, 1, 28033 Madrid, Spain, and provided with Tax Identification Number A-95573317 ("IBERDROLA").

IBERDROLA is represented herein by Mr Javier Salazar Blanco, holder of Spanish identity card number and Mr Gerardo Codes Calatrava, holder of Spanish identity card number both acting in their capacity as joint directors (administradores mancomunados) pursuant to the public deed granted before the Notary Public of Bilbao, Mr José Antonio Isusi Ezcurdia on 23 April 2015 under number 1771 of his official records.

(3) IBERDROLA, S.A., a listed public limited company (sociedad anónima) incorporated in Spain, registered with the Commercial Registry of Vizcaya under volume 17, sheet 114, page number 901 (currently BI-167-A), whose registered office is at Plaza Euskadi, 5, Bilbao (Vizcaya), Spain, and provided with Tax Identification Number A-48010615 ("IBERDROLA PARENT").

IBERDROLA PARENT is represented herein by Mr Pedro Javier Azagra Blázquez, holder of Spanish identity card number and Mr Juan Carlos Rebollo Liceaga, holder of Spanish identity card number acting in their capacity as joint attorneys (apoderados mancomunados) pursuant to the general power of attorney granted before the Notary Public of Madrid, Mr Miguel Ruiz Gallardón García de la Rasilla on 17 May 2016 under number 2228 of his official records.

SIEMENS, IBERDROLA and IBERDROLA PARENT shall be hereinafter referred to as the "Parties" and each of them, individually, as a "Party".

WHEREAS

(A) IBERDROLA, a wholly-owned subsidiary of IBERDROLA PARENT, holds shares in GAMESA CORPORACIÓN TECNOLÓGICA, S.A., a listed public limited company (sociedad anónima) incorporated in Spain, registered with the Commercial Registry of Vizcaya under volume 5147, sheet 7, page number BI-56858, with registered office at Parque Tecnológico de Bizkaia, Edificio 222, 48170 Zamudio (Vizcaya), Spain ("GAMESA"). IBERDROLA's shareholding in GAMESA represents 19.686% of GAMESA's share capital.



- (B) SIEMENS is the direct and indirect owner of companies and assets engaged in (i) the design, development, manufacturing and commissioning of wind power generators, and (ii) the provision of wind turbines' operation and maintenance services ("SIEMENS Wind Power Business").
- (C) Each of SIEMENS and GAMESA have been examining options in order to enhance and maximize the growth opportunities of, respectively, GAMESA and the SIEMENS Wind Power Business, with a view to consolidate and strengthen their respective positions in the renewables energy markets. In this context, they have been discussing and negotiating a potential combination of GAMESA and the SIEMENS Wind Power Business, based on their industrial fit and the synergies that may derive therefrom. Both SIEMENS and IBERDROLA believe that such combination would create a business with a strategic positioning that would be out of reach for each of GAMESA and the SIEMENS Wind Power Business separately. They further believe that such combination would be in the best interest not only of GAMESA and the SIEMENS Wind Power Business but also of their respective shareholders.
- (D) The combination of GAMESA and the SIEMENS Wind Power Business shall be implemented by way of a merger of SIEMENS Wind Power Parent (as defined below) into GAMESA ("Merger" or "Transaction"). To this end, as of the date of this Agreement GAMESA and SIEMENS have entered into a merger agreement governing the framework for the implementation of the Merger ("Merger Agreement").
- (E) In the context of the Transaction the Parties have agreed to enter into this agreement ("Agreement") which shall govern (i) certain rights and obligations of the Parties with respect to the implementation of the Transaction as well as (ii) certain aspects of the relationship between SIEMENS and IBERDROLA in their capacity as future shareholders of GAMESA pursuant to the following:

I. GENERAL

1 Definitions

- **1.1** Terms used in this Agreement shall have the meaning ascribed to them in Schedule 1.1 (Definitions).
- 1.2 The Schedules form part of this Agreement and shall have the same force and effect as if expressly set out in the body of this Agreement.

2 Purpose of this Agreement

- 2.1 The purpose of this Agreement is to set forth:
 - 2.1.1 certain rights and obligations of the Parties with respect to the implementation and other undertakings of the Parties in the context of the Transaction; and
 - 2.1.2 certain aspects of the relationship between SIEMENS and IBERDROLA as future shareholders of GAMESA.
- 2.2 It is IBERDROLA's and SIEMENS' intention to create a leading wind player both in onshore and offshore, creating customer benefits due to (i) a most comprehensive product portfolio, (ii) a large geographical footprint, including a global manufacturing footprint, and (iii) an excellent customer service with one of the widest service networks in the industry. Both



IBERDROLA and SIEMENS agree that such will be the spirit guiding the future collaboration of the Parties, as shareholders of the company, within the framework of this Agreement (and for the avoidance of doubt in accordance with Clause 20.3).

II. TRANSACTION-RELATED UNDERTAKINGS

3 Transaction structure and process

3.1 Transaction structure

- 3.1.1 The Transaction will be structured as a combination of GAMESA's business and the SIEMENS Wind Power Business by way of a statutory merger pursuant to which SIEMENS Wind Power Parent would merge with and into GAMESA.
- 3.1.2 In order to allow for the Merger SIEMENS will implement an internal carve-out process, as a result of which the SIEMENS Wind Power Business shall be held through a Spanish limited flability company (S.L.) ("SIEMENS Wind Power Parent").
- 3.1.3 It is anticipated that, as a consequence of the Merger SIEMENS as sole shareholder of SIEMENS Wind Power Parent, in exchange for its stake in the latter, will be issued shares in GAMESA representing the majority of GAMESA's ordinary share capital.
- 3.1.4 The terms and conditions of the Merger are set forth in further detail in the Merger Agreement as well as in the common terms of merger (proyecto común de fusión), which shall be executed by the Board of Directors of GAMESA and the sole director of SIEMENS Wind Power Parent in accordance with applicable law ("Common Terms of Merger").

3.2 Main Transaction milestones and anticipated timing

For the purposes of implementing the Transaction, SIEMENS and IBERDROLA (in its capacity as shareholder and so far as it is able to procure, with respect to GAMESA) anticipate the following milestones and timetable:

- 3.2.1 June 2016: execution of the Common Terms of Merger by the Board of Directors of GAMESA and the sole director of SIEMENS Wind Power Parent, respectively;
- 3.2.2 July/August 2016: issuance of the independent expert's report on the Common Terms of Merger;
- 3,2,3 September 2016: decision on the Merger by (i) the general shareholders' meeting of GAMESA and (ii) by SIEMENS in its capacity as sole shareholder of SIEMENS Wind Power Parent;
- 3.2.4 December 2016/January 2017; completion of the carve-out process as a result of which the SIEMENS Wind Power Business will be held through SIEMENS Wind Power Parent; and
- 3.2.5 March 2017; effectiveness of the Merger.



4 Interim period

Each of SIEMENS (with respect to SIEMENS Wind Power Parent and the SIEMENS Wind Power Business) and IBERDROLA (in its capacity as shareholder and so far as it is able to procure, with respect to GAMESA) undertake to procure that between the date of this Agreement and the date on which the Merger becomes effective ("Interim Period"), GAMESA, on the one hand, and SIEMENS Wind Power Parent and the SIEMENS Wind Power Business, on the other hand, shall carry on their businesses as a going concern, in the ordinary and usual course as carried on prior to the date of this Agreement and in accordance with business and market practices:

During the Interim Period SIEMENS undertakes to refrain from agreeing or getting engaged, without the prior and express consent of IBERDROLA, in any activities, plans or discussions with the directors or the senior managers of GAMESA, relating to the management of GAMESA, different from activities, plans or discussions concerning the ordinary and day-to-day business and operations that are possible and admissible under applicable laws. In particular, and without limitation, during the Interim Period SIEMENS shall refrain from agreeing, planning or designing with the directors or the senior managers of GAMESA:

- (a) the future organizational chart of the top and key managerial position of GAMESA and/or its Group; or
- (b) any remuneration or retention plan or scheme.

5 Cooperation

- 5.1 General cooperation and support of the Transaction
 - 5.1.1 The Parties shall take any actions reasonably required, and co-operate in all respects with each other, with a view to fulfilling their obligations under this Agreement and to support and successfully and expeditiously complete the Transaction.
 - 5.1.2 Each Party shall as soon as practicable notify the other Party of the occurrence, or non-occurrence, of any event which would be reasonably likely to result in the noncompliance with any of its obligations under this Agreement, or which would otherwise be reasonably likely to have a negative impact on the completion of the Transaction.

5.2 Specific obligations

Notwithstanding Clause 5.1, IBERDROLA further undertakes the following:

5.2.1 Approval of the Merger by GAMESA's shareholders

IBERDROLA, in its capacity as shareholder of GAMESA, shall attend the general shareholders' meeting and, with respect to its entire stake in GAMESA's share capital, vote in favour of the approval of each of the resolutions proposed by GAMESA in the context of the Merger, consistently with, and as provided for in, this Agreement.

5.2.2 IBERDROLA's contractual rights triggered by the Transaction



IBERDROLA undertakes not to (and shall procure that any company within its Group does not) exercise any termination right, arising or derived from any existing agreements between IBERDROLA and GAMESA or any companies of their respective Groups that may be triggered by the change of control of GAMESA resulting from the Transaction, If GAMESA so requests IBERDROLA will expressly and formally waive such rights (or procure that the relevant company within its Group does so).

6 Standstill and lock-up

6.1 Standstill obligations

IBERDROLA and IBERDROLA PARENT agree that:

- 6.1.1 they shall not, and shall procure that the companies of their Group shall not, directly or indirectly, during the Interim Period, be involved in any Prohibited Activity.
- 6.1.2 For the purposes of this Clause 6.1 "Prohibited Activity" means:
 - (a) acquiring or seeking to acquire a direct or indirect interest in the share capital of GAMESA;
 - (b) making a general offer for all or any part of the voting share capital of GAMESA;
 - (c) announcing, or taking any action which would require the announcement of, any proposals for any takeover, merger, consolidation or share exchange or similar transaction involving the securities of GAMESA:
 - (d) taking any step which might give rise to any obligation to make any offer for all or any part of the share capital of GAMESA; and
 - entering into any arrangement with, or advising any person in relation to, any of the foregoing.

6.2 Lock-up

During the Interim Period IBERDROLA shall not dispose of or encumber, or seek to dispose of or encumber (including put options or call options), whether wholly or in part, the shares (or any voting rights inherent thereto) held by it in GAMESA's capital at the date of this Agreement.

7 Exclusivity

7.1 Exclusivity

IBERDROLA and IBERDROLA PARENT agree that, during the Interim Period, they shall, and shall procure that any company of their Group and any of their respective directors, employees or advisers and anyone authorised by or acting with the authority of any of the foregoing shall (as applicable):

7.1.1 sease pursuing any discussions existing at the date of the Agreement with any other person relating to or involving any Third Party Transaction:

MY

- 7.1.2 not solicit, initiate, encourage, negotiate, discuss or otherwise seek to procure (whether directly or indirectly) any enquiries, proposals or approaches from any persons in respect of or in connection with a Third Party Transaction;
- 7.1.3 not provide (whether directly or indirectly) any information or documentation to any person in respect of or in connection with a Third Party Transaction; and
- 7.1.4 not take or allow to be taken any other action which may reasonably be expected to have the effect of encouraging or facilitating any Third Party Transaction, prejudicing the success of the Merger and/or that would otherwise delay, prevent, jeopardize or frustrate the Merger.

7.2 Third Party Transactions

During the Interim Period IBERDROLA and/or IBERDROLA PARENT shall inform SIEMENS immediately if either of them, or any company within their Group, have been approached by a third party in relation to a Third Party Transaction (including informing SIEMENS about the details of any Third Party Transaction proposed or pursued by any such third party).

III. RELATIONSHIP BETWEEN SIEMENS AND IBERDROLA AS SHAREHOLDERS OF GAMESA

8 Corporate governance

Upon the Merger becoming effective, SIEMENS and IBERDROLA undertake, in their capacity as shareholders and so far as they are able to procure, with respect to GAMESA, to support that the general principles and rules set forth below shall apply to the corporate governance of GAMESA.

8.1 Composition of the Board of Directors

- 8.1.1 Upon the Merger becoming effective the board of directors of GAMESA ("Board of Directors") shall initially be composed of 13 members, according to the following:
 - (i) four (4) Independent directors: two of which will be current members of the Board (Ms Gloria Hernandez and Mr Andoni Cendoya) and two who will be proposed by the GAMESA's Nominations Committee (Comision de Nombramientos) from a list to be provided by SIEMENS (for the avoidance of doubt, the latter shall also meet the requirements set forth in section 4 of Article 529 duodecles of the Spanish Companies Act (Ley de Sociedades de Capital) and shall not be under any of the circumstances described in the list provided for in such section 4);
 - (ii) five (5) proprietary directors nominated by SIEMENS;
 - (iii) two (2) proprietary directors nominated by IBERDROLA (Ms Sonsoles Rubio and Mr Francisco Javier Villalba);
 - (iv) the Chief Executive Officer (Consejero Delegado). SIEMENS and IBERDROLA agree to support that the current Chief Executive Officer remains in office. If the current Chief Executive Officer has ceased to be in office, his successor in place at the time of effectiveness of the Merger will be a director who will be appointed by GAMESA's Board of Directors



- subject to the prior approval of SIEMENS, and following the report of the Nominations and Remunerations Committee (Comision de Nombramientos y Retribuciones); and
- (v) the Secretary to the Board of Directors, who will be Mr Carlos Rodríguez-Quiroga, who will remain classified as executive director,
- 8.1.2 IBERDROLA (in its capacity as shareholder and so fat as it is able to procure, with respect to GAMESA) undertakes to procure that the general shareholders' meeting of GAMESA which resolves on the Merger shall also resolve on the appointments, resignations and/or removals of directors required to implement the composition of the Board of Directors described in Clause 8:1.1. The effectiveness of any such resolutions, as the case may be, shall be conditional on the effectiveness of the Merger.

8.2 Future nomination of directors

- 8.2.1 IBERDROLA states its interest, as long as this Agreement remains in force and it maintains a significant shareholding in the share capital of GAMESA, in continuing being represented in the Board of Directors. SIEMENS undertakes that, in its capacity as shareholder, it shall in the future exercise its voting rights so as to allow IBERDROLA to have two proprietary directors until Mr Francisco Javier Villalba ceases as director, which is expected to occur in 2017, and one proprietary director as from Mr Francisco Javier Villalba having ceased as director.
- 8.2.2 When Mr Francisco Javier Villalba ceases as director, he, and his consecutive successors, will be replaced by an independent director who will be proposed by GAMESA's Nominations and Remunerations Committee (Comisión de Nombramientos y Retribuciones) from a list to be provided by IBERDROLA (for the avoidance of doubt such person will meet the requirements set forth in section 4 of Article 529 duodecies of the Spanish Companies Act (Ley de Sociedades de Capital) and shall not be under any of the circumstances described in the list provided for in such section 4).
- 8.2.3 The Parties agree that as long as this Agreement remains in force they undertake to support that the Board of Directors has, at least, four (4) independent directors, and one (1) more once Mr Francisco Javier Villalba has ceased as director.
- 8.2.4 When Ms Gloria Hernández and/or Mr Andoni Cendoya cease as directors, they, and their respective consecutive successors; will be replaced by independent directors who will be proposed by GAMESA's Nominations and Remunerations Committee (Comisión de Nombramientos y Retribuciones) from a list to be provided by IBERDROLA (for the avoidance of doubt such persons will meet the requirements set forth in section 4 of Article 529 duodecies of the Spanish Companies Act (Ley de Sociedades de Capital) and shall not be under any of the circumstances described in the list provided for in such section 4).
- 8.2.5 The Parties further agree, as long as this Agreement remains in force and assuming he will continue performing in an efficient and fair manner his functions, to support that Mr Carlos Rodríguez-Quiroga remains in his position as director and secretary of the Board of Directors. In case of Mr Carlos Rodríguez-Quiroga ceasing to be a director of GAMESA, he, and his consecutive successors, will be replaced by a director who will be proposed by GAMESA's Board of Directors from



a list to be provided by IBERDROLA, following the report of its Nominations and Remunerations Committee (Comisión de Nombramientos y Retribuciones).

8.3 Committees

8.3.1 Audit, Compliance and Related Party Transactions Committee

Upon the Merger becoming effective, GAMESA's Audit and Compliance Committee (Comisión de Auditoria y Cumplimiento) will be composed of three (3) independent directors and renamed to Audit, Compliance and Related Party Transactions Committee (Comisión de Auditoria, Cumplimiento y Operaciones Vinculadas).

As long as this Agreement remains in force, SIEMENS will support that IBERDROLA proposes one of the independent directors forming part of this committee. Additionally, if Ms Gloria Hernández ceases to be a director, IBERDROLA will propose, as replacement, another member who will also be an independent director and SIEMENS will support such proposal. The third member of the committee shall be nominated by SIEMENS.

This committee shall be chaired, during eighteen (18) months as from the effectiveness of the Merger, by Ms Gloria Hernandez, independent director, or, if during said period Ms Gloria Hernandez ceases to be a director, by the independent director proposed by IBERDROLA as replacement. Thereafter, the third member, nominated by SIEMENS, will chair the committee.

The Audit, Compliance and Related Party Transactions Committee shall be in charge, among others, of monitoring any transactions to be entered into by and between, on the one side, GAMESA and, on the other side, SIEMENS and/or IBERDROLA and the companies of their respective Groups.

8.3.2 Nominations and Remunerations Committee

Upon the Merger becoming effective Mr Andoni Cendoya, independent director, shall be member and Chalrman of GAMESA's Nominations and Remunerations Committee (*Comisión de Nombramientos y Retribuciones*), which will be composed of five (5) directors.

Three (3) members of the committee will be proposed by SIEMENS, two of whom will be independent directors.

IBERDROLA will propose one (1) member of this committee and, only if Mr Andoni Cendoya ceases to be a director, an additional member, who will be an independent director and chair the committee.

The Nominations and Remunerations Committee shall also be in charge of submitting to the Audit, Compliance and Related Party Transactions Committee of GAMESA a report, which shall be approved with the favourable vote of all the members of the Nominations and Remunerations Committee, in relation to the proposal for the appointment and removal of the director of the Internal Audit Area of GAMESA.

8.4 Chairman

Upon the Merger becoming effective, the Chairman of the Board of Directors shall be nominated among those proprletary directors nominated by SIEMENS and shall have a casting vote provided that, for the first Chairman and any replacements within a period of



four (4) years as from the Merger becoming effective, when proceeding with the nomination of such Chairman SIEMENS shall (i) consult with IBERDROLA with the intention to use the opinion and experience of IBERDROLA, taking into account its indepth knowledge and experience in the Spanish market and business environment and (ii) nominate a candidate who should speak Spanish and be familiar with the Spanish culture and society. IBERDROLA shall support the nomination without restrictions. As from the fourth anniversary of the effectiveness of the Merger, and in respect of any Chairman to be nominated, only paragraph (i) shall apply.

The Chairman shall be different from the Chief Executive Officer.

8.5 Chief Financial Officer

Upon the Merger becoming effective, and until the end of an integration period of between twelve (12) and a maximum of eighteen (18) months (the "Integration Period"), a member of the SIEMENS' Group pre-Merger as of this date shall be appointed as Chief Financial Officer of GAMESA.

Once the Integration Period has elapsed, and, in any case, not later than eighteen (18) months upon the Merger becoming effective, IBERDROLA shall have the right to propose a candidate for the position of Chief Financial Officer. SIEMENS will support (and procure, so far as it is able to procure, such support at GAMESA) such candidate except if there are material aspects for refusal, in which case IBERDROLA shall subsequently propose further suitable candidates, SIEMENS having the same obligations with respect to such proposals as referred to above.

If the appointed candidate ceases in office (whether voluntarily or involuntarily, including, without limitation, dismissal or removal), IBERDROLA shall have the right to propose a suitable replacement candidate.

The Chief Financial Officer shall discharge his/her office and responsibilities in an adequate and competent manner, including adequately applying and implementing the Group's financial reporting guidelines. Should this not be the case, the Board of Directors, by means of an ordinary resolution, which SIEMENS may support, may remove the Chief Financial Officer from office, if he/she fails to improve the performance within a reasonable cure period. In such case, the Chief Financial Officer will be replaced by a capable candidate having the skills described above to be nominated by IBERDROLA.

8.6 Decisions of GAMESA's Nominations and Remunerations Committee

GAMESA's Nominations and Remunerations Committee, when deliberating and deciding in respect of any appointment or dismissal, including any of those appointments referred to in this Clause 8, may always act freely and under its own criteria, without being bound in any way by any opinion or recommendation whatsoever (for the avoidance of doubt, including any opinion or recommendation of any of the Parties), thus being free to conclude as it deems appropriate. In particular, GAMESA's Nominations and Remunerations Committee shall be free (i) to reject or disregard any or all of the candidates for independent directors that may be proposed in accordance with this Clause 8 and (ii) to propose any other candidates for independent directors it deems appropriate.

8.7 Decisions and sensitive information regarding certain customers

IBERDROLA acknowledges and agrees that after the Merger GAMESA may have certain customers which may be IBERDROLA's competitors and that IBERDROLA's proprietary

Mu

directors will not participate in the commercial decisions in respect of such customers (in accordance with the provisions of article 228.c) of the Spanish Companies Act (Ley de Sociedades de Capital)) and, to the extent legally admissible, will have no access to sensitive information regarding such customers.

8.8 Integration Steering Committee

- 8.8.1 Upon the Merger becoming effective, SIEMENS (in its capacity as major shareholder and so far as it is able to procure with respect to GAMESA) shall procure that GAMESA sets up an integration steering committee for the purpose of steering and supervising the integration of the SIEMENS Wind Power Business with GAMESA ("Integration Steering Committee") during the Integration Period. The Integration Steering Committee will ensure a clear decision process and a fast implementation of the businesses' integration with a focus on delivering the expected value to be created by the Merger.
- 8.8.2 The Integration Steering Committee shall be composed of: the Chief Executive Officer, the Chief Integration Officer (to be appointed by the Board of Directors upon the Merger becoming effective), one member of the Management Board of SIEMENS and one member designated by IBERDROLA. The member nominated by IBERDROLA shall expressly undertake to act exclusively in the interests of GAMESA and not share any sensitive information (in particular, competition-related sensitive information) other than with members of the Board of Directors of GAMESA. Furthermore, the provisions of Clause 8.6 of this Agreement (relating to decisions and information regarding certain customers) shall apply, mutatis mutandis, in the context of the Integration Steering Committee, to the member designated by IBERDROLA.
- 8.8.3 No decision of the Integration Steering Committee shall be taken unless both the Chief Executive Officer and the SIEMENS' member have voted in favour.
- 8.8.4 Each of the relevant integration aspects (such as, among others, operations' optimisation and portfolio rationalisation) shall be led by a dedicated team member of the combined business, such team members to be appointed by the respective corporate and business unit leaders. These dedicated team members shall submit the proposals from the different corporate and business units to the Integration Steering Committee.

9 Headquarters and operations

- 9.1 Corporate seat of GAMESA and functional onshore headquarters SIEMENS undertakes to support that GAMESA's
 - (i) corporate seat under GAMESA's articles of association;
 - (ii) functional and effective headquarters as parent company of its Group; and
 - (iii) functional and effective headquarters for the onshore business remain located in Spain.
- 9.2 Functional offshore headquarters



GAMESA's functional headquarters for the offshore business will be located in Vejle (Denmark) and in Hamburg (Germany).

10 Minority shareholders protection

For the sake of the best interest of the minority shareholders of GAMESA, and considering the best corporate governance practices, the Parties agree that, as long as this Agreement remains in force, the following resolutions shall not be passed, and thus SIEMENS undertakes to vote against the following matters, unless the corresponding resolution proposal has obtained a previous favourable report from the Audit, Compliance and Related Party Transactions Committee or the Nominations and Remunerations Committee of GAMESA, as appropriate, approved by said bodies with the favourable vote of all its members:

- (i) any increase of GAMESA's share capital by more than 10% with the exclusion or limitation of preferential subscription rights;
- (ii) any acquisition, disposal or contribution to another company of assets or activities representing more than 15% of GAMESA's total assets (defined as the sum of all current and non-current assets on the consolidated balance sheet of GAMESA, including the SIEMENS Wind Power Business, as reported in the respective previous fiscal year ("Total Assets")), except intra-Group in favour of subsidiaries that are 100% owned, directly or indirectly, by GAMESA;
- (iii) any merger, demerger or global assignment of assets and liabilities (fusion, escision o cesion global de activos y pasivos), which entails a variation in GAMESA's Total Assets of more than 15%;
- (iv) the winding-up and liquidation of GAMESA; or
- (v) the change in the number of members of GAMESA's Board of Directors.

Without prejudice of the foregoing, the Parties agree that, five (5) Business Days prior to any meeting of the Board of Directors of GAMESA, senior representatives of the Parties shall meet so as to evaluate the content and background of each of the items of the agenda of any such meeting. Any such meetings may also be held by telephone or video conference.

1.1 Squeeze-out; delisting; package sale of shares

11.1 Squeeze-out and/or delisting of GAMESA

If, at any time during a period of eighteen (18) months as from the date of effectiveness of the Merger:

- 11.1.1 SIEMENS formally launches a voluntary takeover bid in Spain in respect of all outstanding shares of GAMESA (i.e. all shares which are, at the relevant date, not yet owned by SIEMENS or by GAMESA) as a result of which a squeeze-out is implemented, resulting in all outstanding shareholders being obliged to sell their shares in GAMESA; or
- 11.1.2 the general shareholders' meeting of GAMESA resolves the de-listing of GAMESA's shares from the official secondary Spanish markets and either GAMESA or SIEMENS formally initiate a de-listing offer (either through a takeover



bld or by means of any other alternative procedure) for all outstanding shares in GAMESA (i.e. all shares which are, at the relevant date, not yet owned by SIEMENS or by GAMESA),

the following shall apply:

- (i) subject to any higher price which may be required to be paid pursuant to the Spanish Takeover Regulations, the minimum per share consideration offered to GAMESA's shareholders shall be EUR 18.15 ("Floor Price"). If after the date of effectiveness of the Merger but before this provision is applied, there occurs (other than where expressly anticipated under this Agreement) either (a) a split or consolidation of GAMESA shares, (b) an extraordinary distribution of dividends or reserves of GAMESA, (c) an increase of capital of GAMESA at a price lower than the market price or (d) any other corporate or financial transaction or other event, in either case having a diluting or concentrative effect on the theoretical value of the GAMESA shares, the Floor Price will be adjusted to ensure that it remains equivalent from a financial perspective; and
- (ii) the per share consideration offered to GAMESA's shareholders shall be entirely in cash, either exclusively or at least as an alternative to any in-kind consideration.

11.2 Package sale of GAMESA shares

As from the date of effectiveness of the Merger and as long as this Agreement remains in force, any transfer, disposal or other form of divestment ("Transfer") of GAMESA shares made by IBERDROLA representing in the aggregate more than 1% (one per cent) of GAMESA's share capital in one or more transactions, to a single third party (or, if they act in a concerted manner, to two or more third parties) ("Transfer Shares"), shall only be permitted pursuant to the provisions of this Clause 11.2:

- 11.2.1 If IBERDROLA intends a Transfer of Transfer Shares it shall first issue a written notice (the "Offer Notice") to SIEMENS containing notification of its bona fide intention to pursue a Transfer of the Transfer Shares.
- 11.2.2 If SIEMENS wishes to buy all of the Transfer Shares it shall, within ten (10) Business Days of the issue of the Offer Notice (the "Offer Closing Date"), send a written notice to IBERDROLA (a "Purchase Notice") containing:
 - (i) an offer to purchase all (and not a specific portion) of the Transfer Shares (the "Offer"); and
 - (ii) the terms on which SIEMENS is prepared to make the Offer, including the price offered for the Transfer Shares.
- 11.2.3 If SIEMENS does not wish to make an Offer it may either send a written notice to IBERDROLA before the Offer Closing Date declining to make an Offer or do nothing in which case it shall not be considered to have made an Offer.
- 11.2.4 Within ten (10) Business Days of receipt of an Offer, IBERDROLA shall send a written notice to SIEMENS indicating whether IBERDROLA accepts the Offer.
- 11.2.5 If IBERDROLA accepts the Offer, the transfer of the relevant Transfer Shares to SIEMENS shall be completed in accordance with the Purchase Notice.
- 11.2.6 If SIEMENS has not sent a Purchase Notice, or declined to make an Offer, by the Offer Closing Date or IBERDROLA has rejected an Offer by SIEMENS in



accordance with Clause 11.2.4, IBERDROLA shall be free to sell any Transfer Shares in respect of which it has not accepted an Offer to any bona fide third party (the "Third Party") within six (6) months of the Offer Closing Date, provided that the price paid by the Third Party is at least as high as the price offered by SIEMENS whose Offer was not accepted, provided that there shall be no minimum price to the extent that, by the Offer Closing Date, SIEMENS has not made Offers or declined making an Offer.

12 Breach

In the event that SIEMENS breaches any of the obligations and undertakings contained in this Section III and in Clause 20.8.3(i)(o)(III) (a "Breach") then any such Breach shall be considered, for the purposes of this Clause, material and IBERDROLA shall have, subject to Clause 12.3, the right to sell the shares in GAMESA that IBERDROLA holds at the time of the Breach (directly or indirectly through majority owned subsidiaries, and provided that the number of the shares that IBERDROLA will be entitled to sell shall not exceed the number of shares it holds in GAMESA on the date hereof) to SIEMENS and SIEMENS shall be obliged to purchase such shares. Subject to Clause 12.5, the purchase price shall be the higher of (a) EUR 22 per share or (b) the closing price for the GAMESA shares on the day the Breach was committed plus a premium of 30%. If the day of the Breach cannot be determined, the price shall be determined on the day of the Breach Notice (as defined below).

Without limiting any of the foregoing, the Parties agree that a Breach shall be deemed to have occurred if SIEMENS (i) fails to comply with the provisions of Clause 8.2 or (ii) fails to exercise its voting rights so as to avoid that a result or situation different from those foreseen in said Clause 8.2 is produced. Such Breach shall be deemed to have occurred despite the occurrence of any of the situations set forth in Clause 8.6. Therefore, and for the purposes hereof, said clause shall not be considered and thus disregarded.

The Parties agree further that, in the cases set forth in sections (ii) and (iii) of Clause 10 above, no Breach shall be deemed to have occurred, unless the 15% threshold is exceeded by more than 5%, calculated over such threshold (i.e., 15%, plus 5% of 15%).

12.2 The purchase by SIEMENS of the relevant shares in accordance with Clause 12.1 shall preclude any other rights of IBERDROLA in respect of the Breach.

12.3 Notification and Cure Period

- 12.3.1 If a Breach occurs, IBERDROLA shall notify SIEMENS in writing specifying the Breach, as soon as reasonably practicable, and requesting SIEMENS to remedy such Breach (the "Breach Notice").
- 12.3.2 The right of IBERDROLA to sell its shares in accordance with Clause 12.1 above will be subject to SIEMENS not having remedied that Breach (or Breaches, as the case may be) and any consequences deriving therefrom within three (3) months following the date of receipt of the Breach Notice (the "Cure Period"). No Cure Period shall be applicable if one specific Breach is relterated within one (1) year after the same Breach was cured.



- 12.4 Where the relevant Breach (or Breaches, as the case may be) are not remedied within the Cure Period and IBERDROLA exercises its right to sell its shares in GAMESA, the transfer of the shares shall take place within ten (10) Business Days as from the lapsing of the Cure Period (the relevant date, being the "Transfer Date").
- 12.5 On the Transfer Date, SIEMENS shall pay to IBERDROLA, in exchange for the transfer of the shares of GAMESA referred to in Clause 12.1, the amount of EUR 22 per share adjusted by any stock split that may have occurred between the date hereof and the date where the Breach at stake occurs (the "Preliminary Purchase Price"). Immediately thereafter, a Valuator shall determine, within seven (7) Business Days as from the Transfer Date; the final purchase price (the "Final Purchase Price"). The Valuator will determine the closing price for the GAMESA shares on the day the Breach was committed plus a premium of 30%. Should such calculation be higher than the Preliminary Purchase Price, then SIEMENS shall pay forthwith, and in any event within five (5) Business Days as from the date on which the Valuator determines the Final Purchase Price, the difference between such valuation and the Preliminary Purchase Price, thus the Final Purchase Price being the sum of the Preliminary Purchase Price plus such difference. For the avoidance of any doubt, where the calculation made by the Valuator leads to an amount which is lower than the Preliminary Purchase Price, no negative adjustment shall be made and the Final Purchase Price shall be the amount of the Preliminary Purchase Price.

IV. MISCELLANEOUS

13 Representations

Each Party represents and warrants to the other Parties that:

- 13.1 it has full power and authority and has taken all actions necessary and obtained all necessary consents to enter into and perform the obligations assumed by it under this Agreement;
- 13.2 its competent corporate bodies have agreed to conclude and execute this Agreement and no further approval or permission is required on its part for concluding and executing this Agreement;
- 13.3 the obligations assumed by it under this Agreement are legal, valid and binding and enforceable against it in accordance with their terms;
- 13.4 its signatories have the right, power and authority to execute this Agreement;
- 13.5 the execution, delivery and performance by it of its obligations under this Agreement do not breach or constitute a default under:
 - (i) any provision of its articles of association or constitutional documents;
 - (ii) any agreement or arrangement to which it (and/or any entity of its Group) is a party or by which it (and/or any entity of its Group) is bound; and
 - (iii) any law, order, judgment or decree of any court or governmental or regulatory authority by which it (and/or any entity of its Group) is bound; and
- 13.6 It is neither insolvent or bankrupt under the laws of its jurisdiction of incorporation, nor unable to pay its debts as they fall due or has proposed or is liable to any arrangement (whether by court process or otherwise) under which its creditors (or any group of them)



would receive less than the amounts due to them; it is not subject to any proceedings in relation to any compromise or arrangement with creditors or any winding up, bankruptcy or insolvency proceedings and no events have occurred which would justify any such proceedings.

- 13.7 Further, SIEMENS represents and warrants to IBERDROLA that, within the framework of the Transaction, the only documents executed, and the only agreements reached, between SIEMENS and GAMESA are (i) the Merger Agreement, its schedules, appendices and the remaining documents referred to therein; and (ii) a *Mutual Options Agreement* entered into between, among others, SIEMENS and GAMESA in respect of Adwen Offshore, S.L. Additionally, SIEMENS represents and undertakes that, during the Interim Period, it will not, without IBERDROLA's prior written consent, execute or enter into any agreement or commitment with GAMESA or its directors or senior managers relating to (a) the management of GAMESA; (b), the future organizational chart of GAMESA and/or its Group; or (c) any remuneration or retention plan or scheme.
- 13.8 Finally, SIEMENS represents and warrants to IBERDROLA that, there is nothing in the Merger Agreement that conflicts or contradicts the contents of this Agreement.

14 Term

This Agreement will enter into force on the date hereof and shall remain in force until termination pursuant to Clause 15.

15 Termination

- 15.1 This Agreement may be terminated at any time:
 - 15.1.1 by SIEMENS or by IBERDROLA and IBERDROLA PARENT, if the Merger Agreement is terminated without the Merger becoming effective;
 - 15.1.2 by SIEMENS, if IBERDROLA and/or IBERDROLA PARENT are in material breach of their undertakings under this Agreement;
 - 15.1.3 by IBERDROLA and IBERDROLA PARENT, if SIEMENS is in material breach of its undertakings under this Agreement; and
 - 15.1.4 by SIEMENS, if IBERDROLA's participation in GAMESA falls below 5% of its outstanding voting share capital.

For the avoidance of doubt, IBERDROLA and IBERDROLA PARENT shall only be entitled to exercise any right to terminate this Agreement jointly.

- 15.2 Notice of any termination must be given in writing within fifteen (15) Business Days after the terminating Party having become aware of the event triggering a termination right. In the event of termination of this Agreement, this Agreement shall have no further effect, without prejudice to the rights of any Party accrued up to termination (including the right of IBERDROLA to sell the GAMESA shares referred to in, and in accordance with, Clause 12 if a Breach had occurred prior to termination) and save for Clauses 1 15, 16, 17, 18 and 20, which shall remain in full force and effect.
- 15.3 Termination of the Agreement pursuant to this Clause 15 shall be without prejudice to the terminating Party's right to pursue any other remedies available to it.



16 Announcements

- 16.1 immediately after the execution of this Agreement:
 - 16.1.1 In order to comply with Article 531 of the Spanish Companies Act (Ley de Sociedades de Capital), IBERDROLA PARENT shall communicate the execution of this Agreement, attaching a copy thereof:
 - to the Spanish Stock Exchange Commission (Comisión Nacional del Mercado de Valores) through a "comunicación de hecho relevante" pursuant to Article 228 of the Spanish Securities Market Act; and
 - (ii) to GAMESA.
 - 16.1.2 SIEMENS, through an "ad hoc Mitteilung" pursuant to section 15, paragraph 1 of the German Securities Trading Act (WpHG), shall inform about the agreements reached with regard to the Transaction and the main terms and conditions of this Agreement, if required.
- 16.2 Promptly following the communications referred to in Clause 16.1.1 and also pursuant to Article 531 of the Spanish Companies Act (Ley de Sociedades de Cepital), IBERDROLA shall file a copy of this Agreement (together with a sworn translation into Spanish) with the Commercial Registry of Vizcaya.
- 16.3 Except for the announcements, communications and filings referred to in Clauses 16.1 and 16.2, no Party shall, without the express prior consent of the other Party, issue any press release, issue any public document or make any public statement or otherwise make any disclosure to any person who is not a party to this Agreement (other than GAMESA), before or after the effectiveness of the Merger, relating to any of the matters provided for or referred to in this Agreement or any ancillary matter. This Clause 16.3 shall not apply to any announcement or disclosure which is permitted under Clause 17 or under the Merger Agreement.

17 Confidentiality

Save as provided herein, no Party shall disclose any information concerning this Agreement. Notwithstanding the foregoing, this Agreement may be disclosed:

- (i) by each of the Parties hereto to its and its Group's legal and financial advisers or employees on a need-to-know basis and on terms that such advisers or employees undertake to comply with the provisions of this Clause in respect of such information as if they were a party to this Agreement;
- (ii) if the disclosure or use is required to vest the full benefit of this Agreement in the Parties;
- (iii) if the disclosure or use is required for the purpose of any arbitral or judicial proceedings arising out of this Agreement or any other agreement entered into under or pursuant to this Agreement;
- (iv) if this Agreement becomes publicly available (other than by breach of the provisions of this Clause);



- (v) otherwise by any Party if required by any applicable law, any government, court or regulatory authority (Including any stock exchange or securities regulator) or body with jurisdiction over such Party (or over any other company within its Group) or stock exchange rules or any binding judgment, order or requirement of any competent authority; or
- (vi) the other Party has given prior written approval to the disclosure or use;

provided that prior to disclosure or use of any information pursuant to Clause 17(v), to the extent legally possible, the Parties shall co-operate, in good faith, in order to agree the timing and content of any such disclosure or use, so far as practicable.

18 SIEMENS Wind Power Parent Shareholders

- 18.1 IBERDROLA and IBERDROLA PARENT acknowledge and accept that:
 - 18.1.1 prior to the effectiveness of the Merger, in addition to SIEMENS, one or more companies within the SIEMENS Group may also be shareholders of SIEMENS Wind Power Parent (each of them, individually, a "SIEMENS Wind Power Parent Shareholder" and, jointly, the "SIEMENS Wind Power Parent Shareholders") and, consequently and pursuant to the Merger, become also shareholders of GAMESA; and
 - 18.1.2 SIEMENS shall be entitled, but not obliged, to partly assign, delegate or otherwise transfer its rights and obligations under this Agreement to any SIEMENS Wind Power Parent Shareholder prior to the effectiveness of the Merger without prior consent of IBERDROLA or IBERDROLA PARENT.
- In the event of partial assignment, delegation or transfer of rights and obligations pursuant to Clause 18.1.2 above, any rights and obligations arising for SIEMENS pursuant to this Agreement will be deemed to be allocated between SIEMENS and each of the relevant SIEMENS Wind Power Parent Shareholders pro rata to their respective stake in SIEMENS Wind Power Parent (or, following the Merger becoming effective, their respective stake in GAMESA), provided that SIEMENS shall guarantee to IBERDROLA and IBERDROLA PARENT, jointly and severally with the SIEMENS Wind Power Parent Shareholders, the performance and observance by any SIEMENS Wind Power Parent Shareholder of its obligations and undertakings under or pursuant to this Agreement. For the avoidance of doubt, in the event of partial assignment, delegation or transfer of rights and obligations pursuant to Clause 18.1.2 above, any references to SIEMENS under this Agreement shall be deemed to be made also to the SIEMENS Wind Power Parent Shareholder(s) if the context so requires.
- 18.3 If SIEMENS does not partly assign, delegate or otherwise transfer its rights and obligations under this Agreement to the relevant SIEMENS Wind Power Parent Shareholder(s), the following shall apply:
 - 18.3.1 such rights and obligations shall remain solely with SIEMENS;
 - 18.3.2 SIEMENS shall continue exercising any rights on its own;
 - 18:3.3 SIEMENS shall either fulfil any obligations on its own or, if and to the extent required, that fulfilment of any obligations under this Agreement requires any action by any SIEMENS Wind Power Parent Shareholder(s), SIEMENS shall procure that

1. W

the required actions are implemented by the relevant SIEMENS Wind Power Parent Shareholder(s) so that the obligations under this Agreement are fulfilled; and

- 18.3.4 it being understood that, towards IBERDROLA and IBERDROLA PARENT, SIEMENS and any SIEMENS Wind Power Parent Shareholder(s) shall always act with a single voice and that:
 - (I) with respect to any obligations and liabilities assumed by SIEMENS pursuant to this Agreement, neither IBERDROLA nor IBERDROLA and IBERDROLA PARENT jointly shall have direct claim against any SIEMENS Wind Power Parent Shareholder(s) but only against SIEMENS; and
 - (ii) with respect to any rights arising for SIEMENS pursuant to this Agreement, IBERDROLA and IBERDROLA PARENT shall be deemed to have fully fulfilled its respective obligations by discharging them to SIEMENS.

19 Shareholders Agreement Steering Committee

The Parties agree that, in order to follow-up on the performance of, and compliance with, this Agreement, a specific committee shall be set up, which shall be composed (i) on the part of SIEMENS, of the Chief Executive Officer, the Member of the Vorstand of SIEMENS responsible for the Energy division and the Head of Mergers and Acquisitions and (ii) on the part of IBERDROLA, of the Chief Executive Officer and the Chief Development Officer for IBERDROLA PARENT.

As long as this Agreement is in force, this committee shall meet twice a year or at the request of any of the Parties. Meetings shall take place in the location mutually agreed between the Parties.

20 Other provisions

20.1 Costs

Each Party shall pay its own costs and expenses incurred in connection with the preparation, negotiation and/or completion of this Agreement.

20.2 Variations and Waivers

20.2.1 Variations

No variation of this Agreement shall be effective unless made in writing and signed by or on behalf of each of the Partles.

20.2.2 No Waiver

- (i) No failure or delay by any Party or time or indulgence given in exercising any remedy or right under or in relation to this Agreement shall operate as a waiver of the same nor shall any single or partial exercise of any remedy or right preclude any further exercise of the same or the exercise of any other remedy or right.
- (ii) No waiver by any Party of any requirement of this Agreement, or of any remedy or right under this Agreement, shall have effect unless given in writing and signed by such Party. No waiver of any particular breach of the



provisions of this Agreement shall operate as a waiver of any repetition of such breach.

20.3 Entire Agreement. Absence of action in concert

- 20.3.1 This Agreement (together with any documents referred to herein or entered into pursuant to this Agreement) contains the entire agreement and understanding of the Parties and supersedes all prior agreements, understandings or arrangements (both oral and written) relating to the subject matter of this Agreement and any such document.
- 20.3.2 Nothing in this Agreement shall be construed as entailing or creating any action in concert (actuación concertada) between the Parties for the purpose of acquiring control of GAMESA or establishing or implementing any common policy as regards the strategy or management of GAMESA or its Group. In particular, save as expressly set out in this Agreement and notwithstanding any obligations of IBERDROLA hereunder:
 - each of IBERDROLA and SIEMENS shall be free to exercise, as it deems convenient or appropriate, any voting and other rights inherent to its stake in GAMESA; and
 - (ii) any proprietary directors nominated by IBERDROLA or SIEMENS in the Board of Directors shall be free to exercise their voting rights as directors of GAMESA as they deem fit and at their entire discretion in relation to all business matters.

20.4 Guarantee from IBERDROLA PARENT. Exercise of rights

- 20.4.1 IBERDROLA PARENT, unconditionally, irrevocably and jointly and severally, guarantees to SIEMENS the due and punctual performance and observance by IBERDROLA of all its obligations, commitments and undertakings under or pursuant to this Agreement.
- 20.4.2 The Parties acknowledge that IBERDROLA PARENT is a Party to this Agreement mainly for the purposes of providing the parent company guarantee set forth in Clause 20.4.1 and to give certain undertakings under this Agreement with respect to itself. Therefore, any rights or claims against SIEMENS pursuant to this Agreement shall be exercised either by IBERDROLA or by IBERDROLA and IBERDROLA PARENT jointly, it being understood that in ne event IBERDROLA and IBERDROLA PARENT shall be entitled to (i) exercise any rights or bring any claims against SIEMENS separately or (ii) if they exercise any rights or bring any claims against SIEMENS jointly, claim/recover more than once with respect to the same matter.

20.5 Amendment of regulations

Upon the Merger becoming effective, SIEMENS and IBERDROLA shall procure, in their capacity as shareholders and so far as they are able to procure, with respect to GAMESA, that the Regulations of the Board of Directors; of the Nominations and Remunerations Committee and of the Audit, Compliance and Related Party Transactions Committee, as appropriate and to the extent legally possible, are amended so as to reflect therein the provisions contained in this Agreement.

20.6 Assignment

Except as set forth in Clause 18, no Party may, without the prior written consent of the other Parties, assign any of its rights or obligations under this Agreement nor grant any security interest over or otherwise transfer the benefit of the whole or any part of this Agreement.

20.7 Counterparts

- 20.7.1 This Agreement may be executed in any number of counterparts, and by each Party on separate counterparts. Each counterpart is an original, but all counterparts shall together constitute one and the same instrument. Delivery of a counterpart of this Agreement by e-mail attachment or fax shall be an effective mode of delivery.
- 20.7.2 In the event the Parties execute this Agreement in counterparts as referred to in Clause 20.7.1, they shall promptly following such execution provide the other Party with signed originals.

20.8 Invalidity and supervening circumstances

- 20.8.1 If any provision in this Agreement shall be held to be illegal, invalid or unenforceable in whole or in part, the provision shall apply with whatever deletion or modification is necessary so that the provision is legal, valid and enforceable and gives effect to the commercial intention of the Parties.
- 20.8.2 To the extent it is not possible to delete or modify the provision, in whole or in part, under Clause 20.8.1, then such provision or part of it shall, to the extent that it is illegal, invalid or unenforceable, be deemed not to form part of this Agreement and the legality, validity and enforceability of the remainder of this Agreement shall, subject to any deletion or modification made under Clause 20.8.1, not be affected.
- 20.8.3 The Parties reciprocally state and acknowledge, that their rights and obligations under this Agreement and, particularly, under Section III, also constitute an essential and decisive factor for its execution and their respective contributions and support to the Transaction as set forth herein.

In this regard, should any binding regulatory Requirement (as defined below) supervene, which conflicts with any rights and/or obligations of any of the Parties under this Agreement (a "Conflict"), the Parties undertake to negotiate in good faith the minimum amendments to this Agreement, so as to making it compatible with the relevant binding regulatory Requirement, always preserving the same underlying equilibrium and spirit ("Good Faith Negotiations").



For the purpose hereof, SIEMENS represents and warrants to IBERDROLA that (i) one of the conditions precedent to which the Merger Agreement is subject, consists of the granting by the CNMV of the exemption referred to under article 8.g) of Royal Decree 1066/2007, of 27 July, on takeover bids, and that, (ii) for the cases where the CNMV indicates that said exemption can only be granted to SIEMENS if (a) certain restrictions or covenants are imposed or (b) certain undertakings are offered (either of (a) or (b) a "Requirement"), pursuant to the terms of the Merger Agreement, SIEMENS shall not be obliged to accept any restrictions or covenants or offer any undertakings that are deemed as unduly burdensome.

- (i) Should the Conflict relate to (1) any of the rights and/or obligations of the Parties under Section III or (2) Clauses 5 or 20.8 of this Agreement, the following provisions shall apply:
 - (a) Good Faith Negotiations shall be conducted during a period of fifteen (15) Business Days as from the date on which any of the Parties first serves a notice to the other Party in order to start negotiations (the "First Negotiation Period").
 - (b) Where, within the First Negotiation Period the Parties reach an agreement regarding the amendments to this Agreement so as to making it compatible with the Requirement, they shall forthwith implement such amendments and SIEMENS shall accept or offer, as applicable, the Requirement.
 - (c) If the Parties fall to reach an agreement within the First Negotiation Period, the following shall apply:
 - (I) Within five (5) Business Days as from the lapsing of the First Negotlation Period (the "First Termination Notification Period"), SIEMENS shall notify to IBERDROLA whether it intends or not to accept or offer, as applicable, the Requirement, Upon this notification having been served, the Chief Executive Officers of both IBERDROLA PARENT and SIEMENS shall meet in order to try to reach an agreement within the period of ten (10) Business Days as from the date on which such notice was served by SIEMENS to IBERDROLA (the "Second Negotiation Period").

If no agreement can be reached within the Second Negotiation Period and SIEMENS decides, irrevocably and in a definitive manner, that it will (i) not accept or offer, as applicable, the Requirement, and that, additionally, it will (ii) not walve the condition precedent to the Merger consisting in the exemption referred to under article 8.g) of Royal Decree 1066/2007, of 27 July, on takeover bids (the "Termination Decision"). SIEMENS shall notify IBERDROLA and GAMESA of the Termination Decision without undue delay.

The Termination Decision shall be definitive and irrevocable and SIEMENS undertakes, and represents and warrants to IBERDROLA, that the termination of the Merger Agreement

and the Merger shall become effective, in any case, (a) on the date on which GAMESA accepts such termination, if such date is prior to the long stop date provided for in the Merger Agreement in respect of the condition precedent referred to above, or (b) on the said long stop date, if GAMESA does not accept the termination prior to such date (either of (a) or (b) the "Merger Termination Date").

- (II) Where, within five (5) Business Days as from lapsing of the Second Negotiation Period (the "Exercise Period"), SIEMENS has not notified the Termination Decision to IBERDROLA or, for whatever reason, the Merger Agreement and the Merger have not been terminated, IBERDROLA shall have the right to request from SIEMENS that, within five (5) Business Days as from the lapsing of the Exercise Period (the "Second Termination Notification Period"), SIEMENS formally notifies GAMESA of its Termination Decision and offers GAMESA to forthwith terminate the Merger Agreement and the Merger, in which case IBERDROLA shall be entitled to terminate this Agreement. In these cases, the termination of the Merger Agreement and the Merger shall take effect on the Merger Termination Date.
- (III) If SIEMENS fails to comply with the request from IBERDROLA pursuant to paragraph (II) (including, for the avoidance of doubt, SIEMENS failing to terminate the Merger Agreement and the Merger if so accepted by GAMESA following receipt of SIEMENS' Termination Decision) within five (5) Business Days as from the lapsing of the Second Termination Notification Period, this shall be considered as a material Breach of SIEMENS for the purpose of Clause 12 and IBERDROLA shall be entitled to exercise its rights under such Clause 12. Said rights shall be enforceable as from the Merger Termination Date.

This paragraph (III) shall also apply, and the rights under Clause 12 shall also be exercisable and enforceable by IBERDROLA as from the Merger Termination Date, if, on such date, SIEMENS has not terminated the Merger Agreement and the Merger and thus, the Merger Agreement and the procedure towards the Merger remain in force following the Merger Termination Date, which shall be deemed as a material Breach.

- (ii) Should the Conflict relate to any of the rights and/or obligations of the Parties under any provision of this Agreement other than those referred to under Clause 20.8.3(l), the following provisions shall apply:
 - (a) The Parties undertake to pursue Good Faith Negotiations so as to remove the Conflict by making the conflicting provisions(s)

- compatible with the relevant Requirement, always preserving the same underlying equilibrium and spirit.
- (b) The effectiveness of the relevant provision(s) under this Agreement causing the Conflict shall be suspended as from the moment in which the Conflict arises and, consequently, there shall be no breach by either Party under this Agreement as long as, and until, the Parties have reached an agreement in accordance with Clause 20.8.3(ii)(a) above. As a result of such suspension the relevant conflict between this Agreement and the relevant Requirement shall be deemed resolved and shall not prevent SIEMENS from completing the Merger under the Merger Agreement.
- (iii) In the event that the duration of the various terms set forth in the previous paragraphs (i) and (ii) is incompatible with any legal terms and/or the term afforded to the CNMV to resolve on the exemption referred to under article 8.g) of Royal Decree 1066/2007, of 27 July, on takeover bids in connection with the Merger, the Parties agree to adapt such terms accordingly.
- (iv) For the avoidance of doubt, SIEMENS' obligations under this Clause 20.8.3 shall be without prejudice to SIEMENS' right to waive the condition precedent to the Merger consisting in the exemption referred to under article 8.g) of Royal Decree 1066/2007, of 27 July, on takeover bids. The Parties agree that, in these cases, if SIEMENS were to file any offer in respect of shares of GAMESA, IBERDROLA shall not be bound by any limitations and shall be free to evaluate, and to decide whether it supports or not, such offer, without any restrictions whatsoever.

20.9 Notices

- 20.9.1 Any notice or other communication in connection with this Agreement (each, a "Notice") shall be:
 - (i) in writing and in English; and
 - (ii) delivered by hand, fax, registered post, e-mail or by courier using an internationally recognised courier company.
- 20.9.2 A Notice to any other Party shall be sent to it at the following address, or such other person or address as the relevant Party may notify to the other Parties from time to time:
 - if sent to SIEMENS:

Addressee: Mr Roland Meinzer (Senior Vice President Post-Closing

Management)

St.-Martin-Str. 76

81541 München

Germany

cc: Mr Anton Steiger (Chief Counsel Corporate and M&A)

Werner-von-Siemens Strasse 50

91052 Erlangen



Germany

E-mail:

roland.meinzer@siemens.com

cc: anton.steiger@siemens.com

(ii) If sent to IBERDROLA and/or IBERDROLA PARENT:

Addressee:

Mr Manuel Toledano

calle Tomás Redendo 1

28033 Madrid

Spain

E-mail:

mtoledano@iberdrola.es

20.9.3 A Party may change its address for Notices provided that it gives the other Parties not less than five (5) days' prior notice in accordance with this clause. Until the end of such notice period, Notices on either address shall remain effective.

20.10 Governing law and jurisdiction

20.10.1 Governing law

This Agreement and any non-contractual obligations arising out of or in connection with it shall be governed by, and interpreted in accordance with, the laws of Spain.

20.10.2 Jurisdiction

Each of the Parties irrevocably agrees that the courts of the city of Madrid are to have exclusive jurisdiction to settle any dispute which may arise out of or in connection with this Agreement and the documents to be entered into pursuant to it and that accordingly any proceedings arising out of or in connection with this Agreement shall be brought in such courts. Each of the Parties irrevocably submits to the jurisdiction of such courts and walves any objection to proceedings in any such court on the ground of venue or on the ground that proceedings have been brought in an inconvenient forum.



In witness whereof the Parties have executed this Agreement on the date stated on the first page. SIGNED by Ms Dagmar Mundani in her capacity as authorised representative (Prokuristin) on behalf of SIEMENS AKTIENGESELLSCHAFT SIGNED by Dr Christian Zentner in his capacity as authorised representative (Prokurist) on behalf of SIEMENS AKTIENGESELLSCHAFT ് ് ം..2 🗒 Mir Javler Salazar Blanco in his capacity as joint director (administrador mancomunado) on behalf of IBERDROLA PARTICIPACIONES, S.A.U. SIGNED by Mr Gerardo Codes Calatrava in his capacity as joint director (administrador mancomunado) no behalf of IBERDROLA PARTICIPACIONES, S.A.U. SIGNED-by-Mf Pedro Javier Azagra Blázquez in his capacity as joint attorney (apoderado mancomunado) on behalf of IBERDROLA, S.A.

SIGNED by Mr Juan Carlos Rebollo Liceaga

on behalf of iBERDROLA, S.A.

in his capacity as joint attorney (apoderado mancomunado):

Schedule 1.1 Definitions

The following words and expressions where used in this Agreement have the meanings given to them below:

- "Agreement" has the meaning given in Recital (E);
- "Board of Directors" has the meaning given in Clause 8.1.1;
- "Breach" has the meaning given in Clause 12.1;
- "Breach Notice" has the meaning given in Clause 12.3.1;
- "Business Day" means any day other than a Saturday, Sunday or bank or public holiday in Spain or in the Federal Republic of Germany;
- "Conflict" has the meaning given in Clause 20.8,3;
- "CNMV" means the Spanish Stock Exchange Commission (Comisión Nacional del Mercado de Valores);
- "Common Terms of Merger" has the meaning given in Clause 3.1.4;
- "Cure Period" has the meaning given in Clause 12.3.2;
- "Exercise Period" has the meaning given in Clause 20.8.3;
- "First Negotiation Period" has the meaning given in Clause 20.8.3;
- "First Termination Notification Period" has the meaning given in Clause 20.8.3;
- "Final Purchase Price" has the meaning given in Clause 12.5;
- *Floor Price" has the meaning given in paragraph (i) of Clause 11.1;
- "GAMESA" has the meaning given in Recital (A);
- "Good Faith Negotiations" has the meaning given in Clause 20.8.3;
- "Group" means with respect to a company, such company and any undertaking which is a subsidiary undertaking of the company from time to time;
- "IBERDROLA" has the meaning given in the Parties' section under (2);
- "IBERDROLA PARENT" has the meaning given in the Parties' section under (3);
- "Integration Period" has the meaning given in Clause 8.5;
- "Integration Steering Committee" has the meaning given in Clause 8.8.1;
- "Interim Period" has the meaning given in Clause 4;
- "Merger" has the meaning given in Recital (D);
- "Merger Agreement" has the meaning given in Recital (D);
- "Merger Termination Date" has the meaning given in Clause 20.8.3;



"Notice" has the meaning given in Clause 20.9.1;

"Offer" has the meaning given in Clause 11.2.2(i);

"Offer Closing Date" has the meaning given in Clause 11.2.2;

"Offer Notice" has the meaning given in Clause 11.2.1;

"Parties" refers to SIEMENS, IBERDROLA and IBERDROLA PARENT and each of them, individually, as a "Party";

"Preliminary Purchase Price" has the meaning given in Clause 12.5;

"Prohibited Activity" has the meaning given in Clause 6.1.2;

"Purchase Notice" has the meaning given in Clause 11.2.2;

"Requirement" has the meaning given in Clause 20.8.3;

"Second Negotiation Period" has the meaning given in Clause 20.8.3;

"Second Termination Notification Period" has the meaning given in Clause 20.8.3;

"SIEMENS" has the meaning given in the Parties' section under (1);

"SIEMENS Wind Power Business" has the meaning given in Recital (B);

"SIEMENS Wind Power Parent" has the meaning given in Clause 3.1.2;

"SIEMENS Wind Power Parent Shareholder" has the meaning given in Clause 18.1.1;

"Spanish Takeover Regulations" means Royal Decree 1066/2007, of 27 July, on Takeover Bids (Real Decreto 1066/2007, sobre el régimen de las ofertas públicas de adquisición de valores), as amended from time to time;

"Termination Decision" has the meaning given in Clause 20.8.3;

"Third Party" has the meaning given in Clause 11.2.6;

"Third Party Transaction" means a proposal in respect of an offer, merger or business combination or similar transaction (whether or not subject to any pre-conditions), proposed by a third party and the purpose of which is to enable that third party (or any other person) to acquire all or a significant proportion of GAMESA's shares or all or a significant proportion of GAMESA's assets or business, or any other arrangement or transaction or series of the same which would be inconsistent with, or detrimental to, the implementation of the Merger;

"Total Assets" has the meaning given in Clause 10(ii);

"Transaction" has the meaning given in Recital (D);

"Transfer" has the meaning given in Clause 11.2;

"Transfer Date" has the meaning given in Clause 12.4;

"Transfer Shares" has the meaning given in Clause 11.2;

40

"Valuator" shall be an international well reputed investment bank jointly selected by the Parties within five (5) Business Days as from lapsing of the Cure Period. Where the Parties are not able to agree on a Valuator within such period, SIEMENS shall chose one out of a list of three (3) international reputed investment banks willing to accept the mandate, proposed by IBERDROLA, and if SIEMENS refuses to make such selection, IBERDROLA shall be entitled to do so.

ACUERDO RELACIONADO CON LA FUSIÓN ENTRE GAMESA Y SIEMENS WIND POWER PARENT

ESTA TRADUCCIÓN TIENE CARÁCTER MERAMENTE INFORMATIVO.

EN CASO DE CUALQUIER DISCREPANCIA ENTRE LA VERSIÓN EN CASTELLANO Y LA VERSIÓN EN INGLÉS, PREVALECERÁ ESTA ÚLTIMA.

17 de junio de 2016

[página 1]

17 de junio de 2016

MARÍA DÁVILA BLANCO Traductora - Intérprete Jurada de INGLÉS Nº 6181

SIEMENS AKTIENGESELLSCHAFT IBERDROLA PARTICIPACIONES, S.A.U. e IBERDROLA, S.A.

ACUERDO RELACIONADO CON LA FUSIÓN ENTRE GAMESA Y SIEMENS WIND POWER PARENT

MARÍA DÁVILA BLANCO

Traductora - Intérprete Jurada de INGLÉS Nº 6181

[página 2]

Índice de Contenidos

Contenido Página			
PARTES1			
CONSIDERANDOS1			
I.	GENERAL2		
1	Definiciones2		
2	Objeto del Acuerdo2		
II.	COMPROMISOS RELACIONADOS CON LA OPERACIÓN3		
3	Estructura y procedimiento de la Operación3		
4	Periodo Interino4		
5	Colaboración4		
6	Prohibiciones de adquisición y disposición5		
7	Exclusividad5		
III.	RELACIÓN ENTRE SIEMENS E IBERDROLA COMO ACCIONISTAS DE GAMESA6		
8	Gobierno corporativo6		
9	Sede y operaciones11		
10	Protección de los accionistas minoritarios11		
11	Squeeze-out (expulsión de minoritarios de la sociedad); exclusión de cotización; venta de		
	paquetes accionariales12		
12	Incumplimiento13		
IV.	VARIOS15		
13	Manifestaciones15		
14	Plazo16		
15	Terminación16		
16	Anuncios17		
17	Confidencialidad17		
18	Socios de SIEMENS Wind Power Parent18		
19	Comité de Seguimiento del Acuerdo de Accionistas19		
20	Disposiciones adicionales19		
Anex	Anexo 1.1 Definiciones29		

[página 3]

En Madrid, a 17 de junio de 2016.

PARTES

(1) SIEMENS AKTIENGESELLSCHAFT, sociedad anónima constituída en Alemania, inscrita en el Registro Mercantil (*Handelsregister*) del juzgado municipal (*Amtsgericht*) de Múnich (Alemania) con el número de registro HRB 6684 y del juzgado municipal (*Amtsgericht*) de Berlin-Charlottenburg (Alemania) con el número de registro HRB 12300, con domicilio social en Wittelsbacherplatz, 2, 80333 Múnich, Alemania ("SIEMENS").

SIEMENS está representada en este acto por Dña. Dagmar Mundani y el Dr. Christian Zentner, actuando mancomunadamente en su condición de apoderados (*Prokuristen*).

(2) IBERDROLA PARTICIPACIONES, S.A.U., sociedad anónima unipersonal, constituida en España e inscrita en el Registro Mercantil de Madrid, al tomo 33515, folio 45, hoja M-603299, con domicilio social en calle Tomás Redondo 1, 28033, Madrid, España y con N.I.F. A-95573317 ("IBERDROLA").

IBERDROLA está representada en este acto por D. Javier Salazar Blanco, titular de documento nacional de identidad español número y D. Gerardo Codes Calatrava, titular de documento nacional de identidad español número ambos actuando en su condición de administradores mancomunados de IBERDROLA, cargo para el que fueron nombrados por virtud de escritura otorgada el 23 de abril de 2015 ante el notario de Bilbao, D. José Antonio Isusi Ezcurdia, con el número 1771 de su protocolo.

(3) IBERDROLA, S.A., sociedad anónima, constituida en España e inscrita en el Registro Mercantil de Vizcaya al tomo 17, folio 114, hoja 901 (actualmente BI-167-A), con domicilio social en Plaza Euskadi, 5, Bilbao (Vizcaya), España y con N.I.F. A-48010615 ("IBERDROLA PARENT").

IBERDROLA PARENT está representada en este acto por D. Pedro Javier Azagra Blázquez, titular de documento de identidad español número y D. Juan Carlos Rebollo Liceaga, titular de documento de identidad español número actuando en su condición de apoderados mancomunados por virtud de escritura de poder otorgada el 17 de mayo de 2016 ante el notario de Madrid, D. Miguel Ruiz-Gallardón García de la Rasilla, con el número 2228 de su protocolo.

En adelante, se hará referencia a SIEMENS, IBERDROLA e IBERDROLA PARENT en conjunto como las "Partes" y a cada una de ellas de forma individual como una "Parte".

CONSIDERANDOS

(A) IBERDROLA, sociedad filial 100% participada por IBERDROLA PARENT, es títular de acciones en GAMESA CORPORACIÓN TECNOLÓGICA, S.A., sociedad anónima cotizada constituida en España e inscrita en el Registro Mercantil de Vizcaya al tomo 5147, página 7, hoja BI-56858, con domicilio social en Parque Tecnológico de Bizkaia, Edificio 222, 48170 Zamudio (Vizcaya), España ("GAMESA"). La participación de IBERDROLA en GAMESA representa un 19,686% del capital social de GAMESA.

[pagina 4]

- (B) SIEMENS es el titular directo e indirecto de ciertas sociedades y activos destinados al (i) diseño, desarrollo, fabricación y puesta en marcha de generadores de energía eólica y (ii) la prestación de servicios de mantenimiento y explotación de turbinas eólicas (el "Negocio Eólico de SIEMENS").
- (C) Tanto SIEMENS como GAMESA han analizado las opciones existentes para mejorar y maximizar las oportunidades de crecimiento de GAMESA y del Negocio Eólico de SIEMENS, respectivamente, a los efectos de consolidar y reforzar sus respectivas posiciones dentro de los mercados de energía renovable. A tal efecto, han negociado y examinado una posible combinación de GAMESA y del Negocio Eólico de SIEMENS teniendo en cuenta la complementariedad industrial y las sinergias que de tal combinación pudieran derivarse. Tanto SIEMENS como IBERDROLA consideran que dicha combinación crearía un negocio con una posición estratégica que hubiera sido imposible alcanzar para el Negocio Eólico de SIEMENS y para GAMESA por separado. Adicionalmente, consideran que dicha combinación supondría un beneficio no solo para GAMESA y el Negocio Eólico de SIEMENS sino también para sus respectivos accionistas.
- (D) La combinación de GAMESA y del Negocio Eólico de SIEMENS se implementará mediante la fusión por la cual SIEMENS Wind Power Parent (según se definirá a continuación) será absorbida por GAMESA (la "Fusión" o la "Operación"). A tal efecto, en la fecha de suscripción del presente Acuerdo, GAMESA y SIEMENS han suscrito un acuerdo de fusión que regulará la implementación de la Fusión (el "Acuerdo de Fusión").
- (E) Dentro del contexto de la Operación, las Partes suscriben el presente acuerdo (el "Acuerdo") que regulará (i) determinadas obligaciones y derechos de las Partes con respecto a la implementación de la Operación así como (ii) determinados aspectos de la relación entre SIEMENS e IBERDROLA en su condición de futuros accionistas de GAMESA de conformidad con lo establecido a continuación.

I. GENERAL

1 Definiciones

- 1.1 Los términos utilizados en este Acuerdo tendrán el significado que se les otorgue en el Anexo 1.1 (Definiciones).
- 1.2 Los Anexos formarán parte integrante del presente Acuerdo y su contenido tendrá la misma validez y efecto que lo dispuesto en el cuerpo del presente Acuerdo.

2 Objeto del Acuerdo

- 2.1 El objeto del presente acuerdo es regular:
 - 2.1.1 determinados derechos y obligaciones de las Partes en relación con la implementación y otros compromisos asumidos dentro del marco de la Operación; y
 - 2.1.2 determinados aspectos de la relación entre SIEMENS e IBERDROLA en su condición de futuros accionistas de GAMESA
- 2.2 IBERDROLA y SIEMENS tienen la intención de crear un operador líder en el sector eólico tanto terrestre como marítimo, que genere beneficios para el cliente gracias a (i) la más completa cartera de productos; (ii) una extensa presencia geográfica, incluida la actividad

[página 5]

de fabricación a nivel global, y (iii) un excelente servicio de atención al cliente con una de/ las redes de servicio más amplias de la industria. Tanto IBERDROLA como SIEMENS están de acuerdo en que éste será el espíritu que guíe la futura colaboración de las Partes, como accionistas de la sociedad, en el marco de este Acuerdo (y, para evitar cualquier duda, de conformidad con lo establecido en la cláusula 20.3).

II. COMPROMISOS RELACIONADOS CON LA OPERACIÓN

3 Estructura y procedimiento de la Operación

3.1 Estructura de la Operación

- 3.1.1 La Operación se estructurará combinando las actividades de GAMESA y el Negocio Eólico de SIEMENS mediante una fusión en virtud de la cual SIEMENS Wind Power Parent será absorbida por GAMESA.
- 3.1.2 Al objeto de facilitar la Fusión, SIEMENS implementará un proceso interno de reorganización (carve-out), como resultado del cual el Negocio Eólico de SIEMENS pasará a ser titularidad de una sociedad de responsabilidad limitada (S.L.) española. ("SIEMENS Wind Power Parent").
- 3.1.3 Se prevé que, como consecuencia de la Fusión, SIEMENS en calidad de accionista único de SIEMENS Wind Power Parent, como contraprestación por su participación en la última, recibirá acciones nuevas de GAMESA que representen la mayoría del capital social ordinario de GAMESA.
- 3.1.4 Los términos y condiciones de la Fusión se establecen más detalladamente en el Acuerdo de Fusión así como en el proyecto común de fusión, el cuál será suscrito por el Consejo de Administración de GAMESA y el administrador único de SIEMENS Wind Power Parent de conformidad con la legislación aplicable (el "Proyecto Común de Fusión").

3.2 Principales hitos de la Operación y calendario previsto

Con el objeto de implementar la Operación, SIEMENS e IBERDROLA (en su condición de accionista y en la medida en que sea posible, respecto de GAMESA) tienen previstos los siguientes hitos y calendario:

- 3.2.1 junio de 2016: formalización del Proyecto Común de Fusión por parte del Consejo de Administración de GAMESA y el administrador único de SIEMENS Wind Power Parent, respectivamente;
- 3.2.2 julio/agosto de 2016: emisión del informe de experto independiente sobre el Proyecto Común de Fusión;
- 3.2.3 septiembre de 2016: aprobación de la Fusión por parte de (i) la junta general de accionistas de GAMESA y (ii) SIEMENS en calidad de accionista único de SIEMENS Wind Power Parent;
- 3.2.4 diciembre de 2016/enero de 2017: realización del proceso interno de reorganización (carve-out) que tendrá como resultado la aportación del Negocio Eólico de SIEMENS a favor de SIEMENS Wind Power Parent; y
- 3.2.5 marzo de 2017: efectividad de la Fusión.

[página 6]

4 Periodo Interino

Tanto SIEMENS (en relación con SIEMENS Wind Power Parent y el Negocio Eólico de SIEMENS) como IBERDROLA (en su condición de accionista y en la medida en que sea posible, respecto de GAMESA) harán que, entre la fecha del presente Acuerdo y la fecha de efectividad de la Fusión (el "**Periodo Interino**"), GAMESA, de una parte, y SIEMENS Wind Power Parent y el Negocio Eólico de SIEMENS, de otra, continúen desarrollando sus negocios como empresas en funcionamiento, conforme al curso ordinario y habitual previo a la suscripción del presente Acuerdo y de conformidad con las prácticas habituales de negocio y mercado.

Durante el Periodo Interino SIEMENS se compromete a abstenerse de acordar o participar, sin el previo y expreso consentimiento de IBERDROLA, en cualesquiera actividades, planes o conversaciones, con los consejeros o altos directivos de GAMESA, relativas a la administración de GAMESA, distintas de actividades, planes o conversaciones relativos al curso ordinario del negocio y a operaciones posibles y admisibles de acuerdo con la legislación aplicable. En particular, y sin carácter limitativo, durante el Periodo Interino SIEMENS se abstendrá de acordar, planear o diseñar junto con los consejeros o altos directivos de GAMESA:

- (a) el futuro organigrama de los altos cargos y puestos directivos clave de GAMESA y/o su Grupo; o
- (b) cualesquiera planes o esquemas de remuneración o retención.

5 Colaboración

- 5.1 Colaboración general y respaldo de la Operación
 - 5.1.1 Las Partes realizarán cuantas acciones sean razonablemente necesarias y colaborarán en todos los sentidos con la otra Parte con el objetivo de cumplir con sus obligaciones derivadas del presente Acuerdo y a los efectos de respaldar y completar de manera satisfactoria y expedita la Operación.
 - 5.1.2 Cada una de las Partes notificará a la otra Parte, tan pronto como le sea posible, el acaecimiento o no de cualquier acontecimiento del que pudiera razonablemente derivarse un incumplimiento de cualquiera de las obligaciones contraídas en virtud de este Acuerdo o que en cualquier otro modo pudieran razonablemente afectar negativamente al cierre de la Operación.

5.2 Obligaciones específicas

Sin perjuicio de lo establecido en la Cláusula 5.1, IBERDROLA adicionalmente se compromete a:

- 5.2.1 Aprobación de la Fusión por parte de los accionistas de GAMESA
 - IBERDROLA, en su condición de accionista de GAMESA, asistirá a la junta general de accionistas y, con respecto a la totalidad de su participación en el capital social de GAMESA, votará a favor de todos los acuerdos propuestos por GAMESA en relación con la Fusión, en los términos y de forma coherente con lo establecido en el presente Acuerdo.
- **5.2.2** Derechos contractuales de IBERDROLA ejercitables como consecuencia de la Operación

[página 7]

IBERDROLA se compromete a no ejercitar ningún derecho de resolución (y hará que ninguna sociedad de su Grupo lo haga), resultante o derivado de cualesquiera contratos existentes entre IBERDROLA y GAMESA o cualesquiera sociedades de sus respectivos Grupos que pudieran ser activados por el cambio de control de GAMESA resultante de la Operación. Si GAMESA así se lo solicita, IBERDROLA renunciará de manera formal y expresa a tales derechos (o hará que la correspondiente sociedad de su Grupo así lo haga).

6 Prohibiciones de adquisición y disposición

6.1 Prohibición de adquisición

IBERDROLA e IBERDROLA PARENT acuerdan que:

- 6.1.1 durante el Periodo Interino no participarán y harán que ninguna de las sociedades del Grupo participe, directa o indirectamente, en ningún tipo de Actividad Prohibida.
- 6.1.2 A los efectos de la presente Cláusula 6.1 se entenderá como "Actividad Prohibida":
 - (a) la adquisición o la pretensión de adquisición de una participación directa o indirecta en el capital social de GAMESA;
 - realizar una oferta general por todo o parte del capital social con derecho de voto de GAMESA;
 - (c) anunciar o emprender una acción que requiera anunciar una propuesta de oferta pública, fusión, consolidación, canje de acciones o cualquier otra operación similar en la que se vean afectados los valores de GAMESA;
 - (d) adoptar medidas que pudieran derivarse en cualquier tipo de obligación de presentar una oferta por parte o la totalidad del capital social de GAMESA;
 y
 - (e) suscribir contratos o prestar asesoramiento a cualquier persona en relación con lo anterior.

6.2 Prohibición de disposición

Durante el Periodo Interino, IBERDROLA no enajenará ni gravará, ni intentará enajenar o gravar (incluyendo opciones de venta o de compra), en parte o en su totalidad, las acciones (o los derechos de voto inherentes a las mismas) que ostente en el capital de GAMESA en la fecha de formalización de este Acuerdo.

7 Exclusividad

7.1 Exclusividad

IBERDROLA e IBERDROLA PARENT acuerdan que durante el Periodo Interino realizarán las siguientes actuaciones, y harán que cualesquiera sociedades del Grupo, sus respectivos consejeros, empleados y cualquier persona autorizada por o con capacidad para obrar por los mismos realicen las siguientes actuaciones, en su caso,:

[página 8]

- 7.1.1 poner fin a cualquier tipo de negociación existente a la fecha del Acuerdo cón cualquier otro individuo en relación con o que implique cualquier Operación con Terceros;
- 7.1.2 no solicitar, instar, fomentar, negociar, analizar o en cualquier otra forma intentar conseguir (directa o indirectamente) ningún tipo de solicitud, propuesta o acercamiento con ninguna persona respecto de o en relación con una Operación con Terceros;
- 7.1.3 no facilitar (directa o indirectamente) ningún tipo de información o documentación a ninguna persona respecto de o en relación con una Operación con Terceros; y
- 7.1.4 no realizar ni permitir que se realicen acciones que quepa prever razonablemente que pudieran fomentar o facilitar cualquier tipo de Operación con Terceros, afectando al éxito de la Fusión y/o que pudieran retrasar, evitar, comprometer o frustrar la Fusión.

7.2 Operaciones con Terceros

Durante el Periodo Interino, IBERDROLA y/o IBERDROLA PARENT informarán a SIEMENS de inmediato en el supuesto de que cualquiera de ellos, o cualquier sociedad de su Grupo, reciban cualquier tipo de contacto en relación con una Operación de Terceros (incluyendo informar a SIEMENS sobre los detalles de cualquier Operación con Terceros propuesta o llevada a cabo por dicho tercero).

III. RELACIÓN ENTRE SIEMENS E IBERDROLA COMO ACCIONISTAS DE GAMESA

8 Gobierno corporativo

Una vez la Fusión sea efectiva, SIEMENS e IBERDROLA se comprometen, en su condición de accionistas y en la medida en que sea posible, respecto de GAMESA, a aplicar los principios generales y reglas establecidos a continuación con respecto al gobierno corporativo de GAMESA.

8.1 Composición del Consejo de Administración

- 8.1.1 Una vez la Fusión sea efectiva, el Consejo de Administración de GAMESA (el "Consejo de Administración") estará inicialmente compuesto por 13 miembros, de conformidad con lo establecido a continuación:
 - (i) cuatro (4) consejeros independientes: dos de los cuales serán miembros actuales del Consejo (Dña. Gloria Hernández y D. Andoni Cendoya) y dos serán propuestos por la Comisión de Nombramientos de GAMESA, de entre los comprendidos en una lista proporcionada por SIEMENS (a efectos aclaratorios, estos últimos también deberán cumplir con los requisitos establecidos en el apartado 4 del artículo 529 duodecies de la Ley de Sociedades de Capital y no se encontrarán incursos en ninguna de las circunstancias descritas en la lista establecida en el citado apartado 4);
 - (ii) cinco (5) consejeros dominicales designados por SIEMENS;
 - (iii) dos (2) consejeros dominicales designados por IBERDROLA (Dña. Sonsoles Rubio y D. Francisco Javier Villalba);

[página 9]

- (iv) el Consejero Delegado. SIEMENS e IBERDROLA acuerdan apoyar que el actual Consejero Delegado se mantenga en su cargo. Si el actual Consejero Delegado hubiere cesado en su cargo, su sustituto en la fecha de eficacia de la Fusión será un consejero que será designado por el Consejo de Administración de GAMESA sujeto a la aprobación previa de SIEMENS, y tras el informe de la Comisión de Nombramientos y Retribuciones; y
- el Secretario del Consejo de Administración, que será D. Carlos Rodríguez-Quiroga, quien seguirá manteniendo la naturaleza de consejero ejecutivo.
- 8.1.2 IBERDROLA (en su condición de accionista y en la medida en que sea posible, en relación con GAMESA) se compromete a hacer que la junta general de accionistas de GAMESA que deba adoptar los acuerdos relativos a la Fusión también adopte los acuerdos correspondientes en relación con el nombramiento, la dimisión y/o cese de consejeros, necesarios para implementar la composición del Consejo de Administración descrita en la Cláusula 8.1.1. La efectividad de tales acuerdos, en su caso, se condicionará a la efectividad de la Fusión.

8.2 Futuros nombramientos de consejeros

- 8.2.1 IBERDROLA manifiesta su interés, mientras este Acuerdo esté en vigor y mientras ésta mantenga una participación significativa en el capital social de GAMESA, en continuar estando representado en el Consejo de Administración. SIEMENS se compromete, en su condición de accionista, a ejercitar sus derechos de voto en el futuro de forma que permita a IBERDROLA mantener dos consejeros dominicales hasta el momento en que D. Francisco Javier Villalba cese en su cargo como miembro del Consejo de Administración, lo que se espera ocurra en 2017, y un consejero dominical una vez D. Francisco Javier Villalba haya cesado en su cargo como consejero.
- 8.2.2 En el momento en que D. Francisco Javier Villalba cese en su cargo como miembro del Consejo de Administración, él, y sus sucesivos sustitutos, serán reemplazados por un consejero independiente, que será propuesto por la Comisión de Nombramientos y Retribuciones de GAMESA de entre los comprendidos en una lista proporcionada por IBERDROLA (a efectos aclaratorios, dicha persona deberá cumplir con los requisitos establecidos en el apartado 4 del artículo 529 duodecies de la Ley de Sociedades de Capital y no se encontrará incursa en ninguna de las circunstancias descritas en la lista establecida en el citado apartado 4).
- 8.2.3 Las Partes acuerdan que, mientras este Acuerdo se mantenga en vigor, se comprometen a apoyar que el Consejo de Administración tenga, por lo menos, cuatro (4) consejeros independientes, y uno (1) adicional una vez D. Francisco Javier Villalba haya cesado en su cargo como consejero.
- 8.2.4 Cuando Dña. Gloria Hernández y/o D. Andoni Cendoya cesen en su cargo como consejeros, ellos, y sus respectivos sucesivos sustitutos, serán reemplazados por consejeros independientes que serán propuestos por la Comisión de Nombramientos y Retribuciones de GAMESA de entre los comprendidos en una lista proporcionada por IBERDROLA (a efectos aclaratorios, estos últimos también deberán cumplir con los requisitos establecidos en el apartado 4 del artículo 529

[página 10]

duodecies de la Ley de Sociedades de Capital y no se encontrarán incursos en ninguna de las circunstancias descritas en la lista establecida en el citado apartado 4).

8.2.5 Las Partes asimismo acuerdan que, mientras este Acuerdo esté en vigor y asumiendo que siga desempeñando sus funciones de forma eficiente y correcta, D. Carlos Rodríguez-Quiroga mantenga su cargo como consejero y secretario del Consejo de Administración. En caso de que D. Carlos Rodríguez-Quiroga cese en su cargo como consejero de GAMESA, él, y sus sucesivos sustitutos, serán reemplazados por un consejero que será propuesto por el Consejo de Administración de GAMESA, de una lista proporcionada por IBERDROLA, tras el correspondiente informe de su Comisión de Nombramientos y Retribuciones.

8.3 Comisiones

8.3.1 Comisión de Auditoría, Cumplimiento y Operaciones Vinculadas

Una vez la Fusión sea efectiva, la Comisión de Auditoría y Cumplimiento de GAMESA estará compuesta por tres (3) consejeros independientes y será redenominada como Comisión de Auditoría, Cumplimiento y Operaciones Vinculadas.

En tanto este Acuerdo se mantenga en vigor, SIEMENS apoyará que IBERDROLA proponga uno de los consejeros independientes como miembro de esta Comisión. Adicionalmente, si Dña. Gloria Hernández cesara en su cargo de consejera, IBERDROLA propondrá, en su lugar, un consejero que también tendrá el de carácter independiente y SIEMENS apoyará dicha propuesta. El tercer miembro de la comisión será designado por SIEMENS.

Esta comisión estará presidida, durante los dieciocho (18) meses siguientes a la fecha de efectividad de la Fusión, por Dña. Gloria Hernández, consejera independiente, o, si durante dicho periodo Dña. Gloria Hernández cesara en su cargo de consejera, por el consejero independiente propuesto por IBERDROLA en su lugar. A partir de la finalización de dicho plazo, el tercer consejero, nombrado por SIEMENS, presidirá la comisión.

La Comisión de Auditoría, Cumplimiento y Operaciones Vinculadas se encargará, entre otros, de supervisar todas aquellas operaciones a llevar a cabo entre GAMESA, de un lado, y SIEMENS y/o IBERDROLA y las sociedades de sus respectivos Grupos, del otro lado.

8.3.2 Comisión de Nombramientos y Retribuciones

Una vez la Fusión sea efectiva, D. Andoni Cendoya, consejero independiente, deberá ser miembro y presidente de la Comisión de Nombramientos y Retribuciones de GAMESA, que estará compuesta de cinco (5) consejeros.

Tres (3) de ellos serán propuestos por SIEMENS, dos de los cuales tendrán el carácter de independientes.

IBERDROLA propondrá un (1) miembro de esta Comisión y solo si D. Andoni Cendoya cesara en su cargo de consejero, IBERDROLA propondrá un consejero adicional que será de carácter independiente y que presidirá la Comisión.

La Comisión de Nombramientos y Retribuciones también se encargará de remitir a la Comisión de Auditoría, Cumplimiento y Operaciones Vinculadas de GAMESA un informe, que deberá ser aprobado con el voto favorable de todos los miembros de la Comisión de

MARÍA DÁVILA BLANCO Traductora - Intérprete Jurada de INGLÉS Nº 6181

[página 11]

Nombramientos y Retribuciones, en relación con la propuesta de nombramiento y cese del director del área de auditoría interna de GAMESA.

8.4 Presidente

Una vez la Fusión sea efectiva, se nombrará al Presidente del Consejo de Administración de entre los consejeros dominicales designados por SIEMENS y ostentará voto de calidad, siempre que, para el primer Presidente y aquellas personas que le pudieran sustituir dentro de un periodo de cuatro (4) años contados a partir de la fecha en que la Fusión sea efectiva, cuando se proceda al nombramiento de dicho Presidente, SIEMENS (i) consulte con IBERDROLA con la intención de utilizar la opinión y experiencia de IBERDROLA, tomando en consideración su profundo conocimiento y experiencia en el mercado español y el entorno empresarial y (ii) designe a un candidato que deberá hablar español y conocer la cultura y sociedad españolas. IBERDROLA apoyará su nombramiento sin restricciones. A partir del cuarto aniversario de la fecha de efectividad de la Fusión, y en relación con cualquier Presidente que vaya a ser nombrado, sólo será de aplicación el apartado (i) anterior.

El Presidente será una persona distinta del Consejero Delegado.

8.5 Director Financiero

Una vez la Fusión sea efectiva y hasta la finalización de un periodo de integración de entre doce (12) y un máximo de dieciocho (18) meses (el "**Periodo de Integración**"), un miembro del Grupo SIEMENS previo a la Fusión, que sea miembro del mismo a la fecha del presente Acuerdo, será designado Director Financiero de GAMESA.

Una vez el Periodo de Integración haya transcurrido y, en cualquier caso, no más tarde del transcurso de los dieciocho (18) meses siguientes a la fecha en que la Fusión sea efectiva, IBERDROLA tendrá derecho a proponer un candidato para el cargo de Director Financiero. SIEMENS apoyará a ese candidato (y conseguirá dicho apoyo en GAMESA en la medida en que le sea posible) excepto si hubiere razones significativas para recusarlo, en cuyo caso IBERDROLA deberá proponer sucesivos candidatos adecuados, teniendo SIEMENS las mismas obligaciones anteriormente referidas con respecto a ellos.

Si el candidato nombrado cesa en su cargo (sea de forma voluntaria o involuntaria, incluyendo, sin_carácter_limitativo, dimisión_o_cese), IBERDROLA_tendrá_derecho_a proponer un candidato adecuado que le sustituya.

El Director Financiero deberá delegar su cargo y responsabilidades de manera adecuada y competente, incluyendo una aplicación e implementación adecuadas de las directrices de información financiera del Grupo. En caso contrario, el Consejo de Administración, por medio de un acuerdo ordinario, que SIEMENS podrá apoyar, podrá cesar al Director Financiero de su cargo, si éste no lograra mejorar su desempeño dentro de un plazo de subsanación razonable. En este caso, el Director Financiero será sustituido por un candidato capaz, que posea las aptitudes descritas más arriba, que será designado por IBERDROLA.

8.6 Decisiones de la Comisión de Nombramientos y Retribuciones de GAMESA

La Comisión de Nombramientos y Retribuciones de GAMESA, cuando delibere y tome decisiones relativas a cualesquiera nombramientos o ceses, incluyendo aquellos nombramientos referidos en esta Cláusula 8, podrá actuar libremente en todo momento y de acuerdo con sus propios criterios, sin estar vinculada por ninguna opinión o

[página 12]

recomendación en ningún caso (a efectos aclaratorios, incluyendo cualquier opinión o recomendación de cualquiera de las Partes), siendo por tanto libre de adoptar decisiones según estime apropiado. En concreto, la Comisión de Nombramientos y Retribuciones de GAMESA será libre para (i) rechazar o desestimar a cualesquiera o a todos los candidatos a consejeros independientes que pudieran ser propuestos de acuerdo con lo establecido en esta Cláusula 8 y (ii) proponer cualesquiera otros candidatos a consejeros independientes que estime apropiados.

8.7 Acuerdos e información sensible relativa a ciertos clientes

IBERDROLA reconoce y acuerda que después de la Fusión GAMESA podrá tener ciertos clientes que pudieran ser competidores de IBERDROLA y que los consejeros dominicales de IBERDROLA no participarán en las decisiones comerciales con respecto a dichos clientes (de conformidad con lo dispuesto en el artículo 228.c) de la Ley de Sociedades de Capital) y, en la medida de lo legalmente admisible, no tendrán acceso a información sensible relativa a estos clientes.

- 8.8 Comité de Seguimiento de la Integración
- 8.8.1 Una vez la Fusión sea efectiva, SIEMENS hará (en su condición de accionista mayoritario y en la medida en que pueda hacerlo con respecto a GAMESA) que GAMESA constituya un Comité de Seguimiento de la Integración con el fin de dirigir y supervisar la integración del Negocio Eólico de SIEMENS con GAMESA (el "Comité de Seguimiento de la Integración") durante el Periodo de Integración. El Comité de Seguimiento de la Integración garantizará un proceso de toma de decisiones definido y una rápida implementación de la integración del negocio con el objetivo de generar el valor que se espera crear con la Fusión.
- 8.8.2 El Comité de Seguimiento de la Integración estará formado por: el Consejero Delegado, el Director de Integración (que será nombrado por el Consejo de Administración una vez la Fusión sea efectiva), un miembro del Consejo de Administración (Management Board) de SIEMENS y un miembro designado por IBERDROLA. El miembro nombrado por IBERDROLA deberá comprometerse expresamente a actuar exclusivamente de acuerdo con los intereses de GAMESA y a no revelar ninguna información sensible (en concreto, información sensible relativa a cuestiones de competencia) a cualquier persona distinta de los miembros del Consejo de Administración de GAMESA. Adicionalmente, las provisiones establecidas en la Cláusula 8.6 de este Acuerdo (relacionadas con decisiones e información relativas a determinados clientes) se aplicarán, mutatis mutandis, en el contexto del Comité de Seguimiento de la Integración, al miembro designado por IBERDROLA.
- 8.8.3 Ninguna decisión del Comité de Seguimiento de la Integración podrá ser tomada sin el voto a favor tanto del Consejero Delegado como del miembro de SIEMENS.
- 8.8.4 Cada uno de los aspectos de integración relevantes (tales como, entre otros, optimización de operaciones y racionalización de la cartera) será liderado por un miembro especializado del equipo del negocio combinado, debiendo ser designados dichos miembros del equipo por los directores de la unidad corporativa o de negocio que corresponda. Estos miembros especializados del equipo deberán remitir las propuestas de las distintas unidades corporativas o de negocio al Comité de Seguimiento de la Integración.

[página 13]

9 Sede y operaciones

9.1 Domicilio social de GAMESA y sede operativa del negocio onshore

SIEMENS se compromete a apoyar que

- (i) el domicilio social de GAMESA según sus Estatutos sociales;
- (ii) la sede efectiva y operativa como sociedad matriz de su Grupo; y
- (iii) la sede efectiva y operativa del negocio onshore

permanezcan localizados en España.

9.2 Sede operativa del negocio offshore

La sede operativa del negocio offshore de GAMESA estará localizada en Vejle (Dinamarca) y en Hamburgo (Alemania).

10 Protección de los accionistas minoritarios

De cara a tutelar los intereses de los accionistas minoritarios de GAMESA, y de acuerdo con las buenas prácticas del gobierno corporativo, las Partes acuerdan que, mientras este Acuerdo se mantenga en vigor, no se adoptarán los siguientes acuerdos, y, por tanto, SIEMENS se compromete a votar en contra de los siguientes asuntos, a menos que la propuesta de acuerdo correspondiente hubiera obtenido el informe previo favorable de la Comisión de Auditoría, Cumplimiento y Operaciones Vinculadas o de la Comisión de Nombramientos y Retribuciones de GAMESA, según corresponda, aprobados por dichos órganos con el voto favorable de todos sus miembros:

- (i) cualquier aumento del capital social de GAMESA en más del 10% con exclusión o limitación de derechos de suscripción preferente;
- (ii) cualquier adquisición, enajenación o aportación a otra sociedad de activos o actividades que representen más del 15% de los activos totales de GAMESA (definidos como la suma de todos los activos corrientes y no corrientes que figuren en el balance consolidado de GAMESA, incluyendo el Negocio Eólico de SIEMENS, tal y como se formule en el correspondiente ejercicio social anterior (los "Activos Totales")), excepto las intra-Grupo realizadas a favor de filiales participadas al 100%, de forma directa o indirecta, por GAMESA;
- (iii) cualquier fusión, escisión o cesión global de activos y pasivos que implique una variación en los Activos Totales de GAMESA superior al 15%;
- (iv) la disolución y liquidación de GAMESA; o
- (v) el cambio en el número de miembros del Consejo de Administración de GAMESA.

Sin perjuicio de lo anterior, las Partes acuerdan que, con una antelación de cinco (5) Días Hábiles a cualquier reunión del Consejo de Administración de GAMESA, altos representantes de las Partes se reunirán con el fin de evaluar el contenido y contexto de cada uno de los puntos del orden del día de la reunión correspondiente. Dichas reuniones podrán, asimismo, celebrarse por teléfono o videoconferencia.

[página 14]

- 11 Squeeze-out (exclusión de minoritarios de la sociedad); exclusión de cotización; venta de paquetes accionariales
- 11.1 Squeeze-out y/o exclusión de cotización de GAMESA

En el supuesto de que en cualquier momento durante un periodo de dieciocho (18) meses a partir de la fecha de efectividad de la Fusión:

- 11.1.1 SIEMENS formulase formalmente una oferta pública voluntaria de adquisición en España sobre las acciones en circulación de GAMESA (es decir, las acciones que en tal fecha no fueran titularidad de SIEMENS o GAMESA) y como resultado de tal oferta se implemente un supuesto de squeeze-out que conlleve una obligación para el resto de accionistas de GAMESA a vender sus acciones; o
- 11.1.2 la junta general de accionistas de GAMESA acuerde la exclusión de cotización de las acciones de GAMESA del mercado secundario oficial de España y tanto GAMESA como SIEMENS inicien formalmente una oferta de exclusión (mediante una oferta pública de adquisición o cualquier otro procedimiento alternativo) en relación con todas las acciones en circulación de GAMESA (es decir las acciones que en tal fecha no fueran titularidad de SIEMENS o GAMESA).

será de aplicación lo dispuesto a continuación:

- (i) sujeto a cualquier incremento en el precio que pudiera ser de aplicación según la normativa española sobre ofertas públicas de adquisición de valores, el importe mínimo por acción ofertado a los accionistas de GAMESA será de 18,15 euros ("Precio Mínimo"). En el supuesto de que con posterioridad a la fecha de efectividad de la Fusión pero con anterioridad a la aplicación de la presente disposición suceda (cualquier otro supuesto a excepción de lo regulado expresamente en este Acuerdo) (a) una escisión o consolidación de las acciones de GAMESA, (b) una distribución extraordinaria de dividendos o reservas de GAMESA, (c) un aumento de capital de GAMESA a un precio inferior al precio de mercado o (d) cualquier otra operación financiera o corporativa o cualquier otro supuesto que pudiera suponer un efecto de dilución o concentración en el valor teórico de las acciones de GAMESA, el Precio Mínimo se ajustará de manera que resulte equivalente desde una perspectiva financiera; y
- (ii) la contraprestación que los accionistas de GAMESA perciban por sus acciones se realizará totalmente en efectivo, de manera exclusiva o al menos como alternativa a cualquier tipo de contraprestación en especie.
- 11.2 Venta de paquetes accionariales de GAMESA

A partir de la fecha de efectividad de la Fusión y en la medida en que el presente Acuerdo siga en vigor, cualquier transmisión, enajenación o cualquier otra forma de desinversión (la "Transmisión") de las acciones de GAMESA realizada por IBERDROLA y que represente en conjunto más de un 1% (uno por ciento) del capital social de GAMESA en una o más operaciones, a favor de un único tercero (o, si actuaran de forma conjunta, dos o más terceros) (las "Acciones sujetas a Transmisión"), solamente se permitirá conforme a lo dispuesto en la presente Cláusula 11.2.

11.2.1 En el supuesto de que IBERDROLA pretenda realizar una Transmisión de las Acciones sujetas a Transmisión deberá emitir previamente una comunicación por escrito (la "Comunicación de la Oferta") a SIEMENS que incluya la notificación

[página 15]

de su intención de realizar, de buena fe, una Transmisión de las Acciones sujetas a Transmisión.

- 11.2.2 En el supuesto de que SIEMENS pretenda adquirir todas las Acciones sujetas a Transmisión, deberá, dentro de los diez (10) Días Hábiles siguientes a contar a partir de la Comunicación de la Oferta (la "Fecha de Cierre de la Oferta"), enviar una notificación por escrito a IBERDROLA (la "Comunicación de Compra") que incluya:
 - (i) la oferta de compra de la totalidad (y no de una parte concreta) de las Acciones sujetas a Transmisión (la "**Oferta**"); y
 - (ii) los términos que SIEMENS pretende aplicar a dicha oferta, incluido el precio ofertado por las Acciones sujetas a Transmisión.
- 11.2.3 En el supuesto de que SIEMENS no desee realizar una Oferta, podrá enviar una notificación por escrito a IBERDROLA con anterioridad a la Fecha de Cierre de la Oferta en la que decline realizar una Oferta o no realizará gestión alguna en cuyo caso no se estimará que no ha realizado Oferta alguna.
- 11.2.4 Dentro de los diez (10) Días Hábiles siguientes a contar a partir de la recepción de una Oferta, IBERDROLA enviará una notificación por escrito a SIEMENS en la que se indique si IBERDROLA acepta la Oferta.
- 11.2.5 En el supuesto de que IBERDROLA acepte la Oferta, la transmisión de las correspondientes Acciones sujetas a Transmisión a SIEMENS se completará de conformidad con la Comunicación de Compra.
- 11.2.6 En el supuesto de que SIEMENS no envíe Comunicación de Compra, o decline realizar oferta alguna, en la Fecha de Cierre de la Oferta o IBERDROLA rechace la Oferta realizada por SIEMENS según lo dispuesto en la Cláusula 11.2.4, IBERDROLA podrá libremente vender las Acciones sujetas a Transmisión sobre las cuales no hubiera aceptado Oferta alguna a cualquier tercero ("Tercero") de buena fe dentro de los seis (6) meses siguientes a partir de la Fecha de Cierre de la Oferta, siempre que el precio que deba satisfacer ese Tercero alcance, al menos, el precio ofertado por parte de SIEMENS cuya Oferta no se aceptó, que no existirá un precio mínimo y en la medida en que a la Fecha de Cierre de la Oferta, SIEMENS no hubiera hecho Oferta alguna o hubiera declinado realizar una Oferta.

12 Incumplimiento

En el supuesto de que SIEMENS incumpla alguna de las obligaciones y compromisos contenidos en esta Sección III y en la Cláusula 20.8.3.(i)(c)(III) (un "Incumplimiento"), cualquier Incumplimiento se considerará, a los efectos de lo establecido en esta Cláusula, esencial, e IBERDROLA dispondrá, sujeto a lo establecido en la Cláusula 12.3, del derecho a vender las acciones de GAMESA de las que IBERDROLA sea titular en el momento del Incumplimiento (tanto de forma directa como indirecta, a través de filiales mayoritariamente por ella participadas y teniendo en cuenta que el número de acciones que IBERDROLA tendrá derecho a vender no excederá del número de acciones de GAMESA de las que ésta es titular a la fecha de este Acuerdo) a SIEMENS y SIEMENS estará obligada a adquirir dichas acciones. Sujeto a lo establecido en la Cláusula 12.5, el precio de adquisición será el mayor de (a) 22 euros por acción o (b) el precio de cotización al cierre del mercado de las acciones de GAMESA en el día en que se produjo el

[página 16]

Incumplimiento más una prima del 30%. Si el día del Incumplimiento no se puede determinar, el precio será determinado el día de la Notificación del Incumplimiento (según esta se define más abajo).

Sin que suponga una limitación a lo anterior, las Partes acuerdan que se entenderá que ha ocurrido un Incumplimiento si SIEMENS (i) no cumple con las estipulaciones de la Cláusula 8.2 o (ii) no ejercita sus derechos de voto en el sentido necesario para evitar que se produzca un resultado o situación distintos de los establecidos en dicha Cláusula 8.2. Dicho Incumplimiento se entenderá ocurrido a pesar de que tenga lugar cualquiera de las situaciones descritas en la Cláusula 8.6. Por tanto, y a estos efectos, dicha cláusula no será tenida en consideración.

Las Partes acuerdan, además, que en los supuestos previstos en los apartados (ii) y (iii) de la Cláusula 10 anterior, se entenderá que no ha habido Incumplimiento salvo que el umbral del 15% sea excedido en más de un 5% calculado sobre ese mismo umbral (esto es, 15% más un 5% del 15%).

- 12.2 La adquisición por parte de SIEMENS de las acciones correspondientes de acuerdo con lo establecido en la Cláusula 12.1, impedirá el ejercicio por parte de IBERDROLA de cualesquiera otros derechos en relación con el Incumplimiento.
- 12.3 Notificación y Plazo de Subsanación
 - 12.3.1 Si se produjera un Incumplimiento, IBERDROLA deberá enviar una notificación a SIEMENS por escrito, a la mayor brevedad posible, especificando el Incumplimiento, y solicitando a SIEMENS que subsane el Incumplimiento (la "Notificación del Incumplimiento").
 - 12.3.2 El derecho de IBERDROLA de vender sus acciones, de acuerdo con lo establecido en la Cláusula 12.1 anterior, estará condicionado a que SIEMENS no hubiese subsanado dicho Incumplimiento (o Incumplimientos, en su caso) así como cualesquiera consecuencias resultantes de dicho Incumplimiento, dentro de los tres (3) meses siguientes a la fecha de recepción de la Notificación de Incumplimiento (el "Plazo de Subsanación"). No será aplicable el Plazo de Subsanación en el supuesto de que un Incumplimiento en concreto se repitiese dentro de un plazo de un (1) año contado a partir de la subsanación de ese mismo Incumplimiento.
- 12.4 En caso de que un Incumplimiento (o Incumplimientos, en su caso) no se hubiese subsanado dentro del Plazo de Subsanación e IBERDROLA ejercitase su derecho a vender sus acciones de GAMESA, la transmisión de las acciones deberá producirse dentro de los diez (10) Días Hábiles siguientes a la finalización del Plazo de Subsanación (siendo la fecha correspondiente, la "Fecha de Transmisión").
- 12.5 En la Fecha de Transmisión, SIEMENS abonará a IBERDROLA, como contraprestación por la transmisión de las acciones de GAMESA referida en la Cláusula 12.1, la cantidad de 22 euros por acción ajustada en función de cualquier desdoblamiento de las acciones que hubiera tenido lugar entre la fecha del presente Acuerdo y aquella en que ocurra el correspondiente Incumplimiento (el "Precio de Adquisición Provisional"). Con carácter inmediatamente posterior, un Valorador determinará, dentro de los siete (7) Días Hábiles siguientes a la Fecha de Transmisión, el precio de adquisición final (el "Precio de Adquisición Final"). A estos efectos, el Valorador determinará el precio de cotización al cierre del mercado de las acciones de GAMESA en el día en que se produjo el

[página 17]

Incumplimiento más una prima del 30%. En caso de que tal cálculo resultase en una cantidad superior al Precio de Adquisición Provisional, SIEMENS deberá abonar inmediatamente y, en cualquier caso, dentro de los cinco (5) Días Hábiles siguientes a la fecha en la que el Valorador determine el Precio de Adquisición Final, la diferencia entre dicha valoración y el Precio de Adquisición Provisional, resultando, por tanto, el Precio de Adquisición Final como la suma del Precio de Adquisición Provisional más dicha diferencia. A efectos aclaratorios, en caso de que la valoración realizada por el Valorador resulte en una cantidad inferior al Precio de Adquisición Provisional, no se realizará ajuste negativo alguno, y el Precio de Adquisición Final será el importe del Precio de Adquisición Provisional.

IV. VARIOS

13 Manifestaciones

Cada una de las Partes manifiesta y garantiza a la otra Parte que:

- 13.1 cuenta con facultades y autoridad suficientes y ha emprendido todas las acciones necesarias y obtenido los consentimientos necesarios para formalizar y cumplir las obligaciones asumidas de conformidad con este Acuerdo;
- 13.2 su respectivos órganos sociales han acordado la suscripción y formalización del presente Acuerdo y no son necesarias aprobaciones o permisos adicionales por su parte a los efectos de suscribir y formalizar el presente Acuerdo;
- 13.3 las obligaciones que asume en virtud de este Acuerdo son conformes a la ley, válidas, vinculantes y exigibles frente a tal sociedad de conformidad con sus términos;
- 13.4 sus firmantes tienen capacidad, derechos y facultades para suscribir este Acuerdo;
- **13.5** la suscripción, formalización y cumplimiento de sus obligaciones no implican incumplimiento alguno o violación de:
 - (i) lo dispuesto en sus estatutos sociales o documentos de constitución;
 - (ii) cualesquiera acuerdos o contratos de los que sea parte (y/o cualquiera de las entidades de su Grupo) o que en algún modo le vincule (y/o a cualquiera de las entidades de su Grupo); y
 - (iii) cualquier ley, auto, sentencia o decreto vinculante para la sociedad (y/o cualquier entidad del Grupo) proveniente de cualquier tribunal o autoridad gubernamental o regulatoria; y
- 13.6 no se encuentra en situación de concurso de acreedores o bancarrota de conformidad con la legislación aplicable según su constitución, ni es incapaz de satisfacer sus deudas según se devenguen o haya celebrado o se encuentra vinculada por algún tipo de acuerdo (ya sea derivado de un procedimiento judicial o de cualquier otro tipo) frente a sus acreedores (o cualquier agrupación de los mismos) que implique que éstos reciban una cantidad inferior a la adeudada; no se encuentra sometida a ningún tipo de procedimiento relacionado con acuerdos o compromisos con los acreedores o en un proceso de liquidación, bancarrota o concurso de acreedores y no existen supuestos que pudieran justificar el inicio de tales procedimientos.

Traductora - Intérprete Jurada de MGLÉS Nº 6181

[página 18]

- Asimismo, SIEMENS manifiesta y garantiza a IBERDROLA que, en el marco de lá Operación, los únicos documentos suscritos, y los únicos acuerdos alcanzados entre SIEMENS y GAMESA son (i) el Acuerdo de Fusión, sus anexos, apéndices y los restantes documentos en él referidos; (ii) un Acuerdo de Opciones Recíprocas (*Mutual Options Agreement*) suscrito, entre otros, por SIEMENS y GAMESA con respecto a Adwen Offshore, S.L. Adicionalmente, SIEMENS manifiesta y se compromete a que, durante el Periodo Interino, no ejecutará ni suscribirá ningún acuerdo o compromiso con GAMESA o sus consejeros o altos directivos sin el consentimiento previo y por escrito de IBERDROLA en relación con (a) la administración de GAMESA; (b) el futuro organigrama de GAMESA y/o su Grupo; o (c) cualquier plan o esquema de retribuciones o retenciones.
- **13.8** Finalmente, SIEMENS manifiesta y garantiza a IBERDROLA que no hay nada en el Acuerdo de Fusión que entre en conflicto o contradiga el contenido de este Acuerdo.

14 Plazo

El presente Acuerdo entrará en vigor en la fecha establecida en el mismo y deberá permanecer en vigor hasta su terminación de conformidad con la Cláusula 15.

15 Terminación

- **15.1** El presente Acuerdo podrá ser resuelto en cualquier momento:
 - **15.1.1** por parte de SIEMENS o por parte de IBERDROLA e IBERDROLA PARENT, en el supuesto de que el Acuerdo de Fusión se finalice sin que la Fusión sea efectiva;
 - 15.1.2 por parte de SIEMENS, en el supuesto de que IBERDROLA y/o IBERDROLA PARENT incumplieran sustancialmente sus obligaciones de conformidad con este Acuerdo;
 - 15.1.3 por parte de IBERDROLA e IBERDROLA PARENT, en el supuesto de que SIEMENS incumpliera sustancialmente sus obligaciones de conformidad con este Acuerdo; y
 - 15.1.4 por parte de SIEMENS, en el supuesto de que la participación de IBERDROLA en GAMESA se redujera a un porcentaje inferior al 5% de las acciones con derecho a vote.

A efectos aclaratorios, IBERDROLA e IBERDROLA PARENT solamente podrán ejercer su derecho a resolver el Acuerdo de forma conjunta.

- 15.2 La notificación de resolución del Acuerdo deberá realizarse por escrito dentro de los quince (15) Días Hábiles siguientes a la fecha en que la Parte que lo resuelva hubiera tenido conocimiento del hecho que desencadene su derecho a resolverlo. En el supuesto de que el presente Acuerdo se resuelva, dejará de surtir efecto, sin perjuicio de los derechos que cualquier Parte hubiera devengado hasta la resolución (incluyendo el derecho de IBERDROLA a vender las acciones de GAMESA referido en, y de acuerdo con, la Cláusula 12 si hubiera tenido lugar un Incumplimiento antes de la resolución) y a excepción de las Cláusulas 1, 15, 16, 17, 18 y 20 que permanecerán en vigor y seguirán siendo efectivas.
- 15.3 La terminación del Acuerdo según lo dispuesto en la presente Cláusula 15 no limitará el derecho de las Parte que inste su resolución de ejercitar cualesquiera otras acciones de las que disponga.

[página 19]

16 Anuncios

- **16.1** Con efecto inmediato tras la formalización de este Acuerdo:
 - 16.1.1 Al objeto de cumplir con el artículo 531 de la Ley de Sociedades de Capital, IBERDROLA PARENT comunicará la formalización del presente Acuerdo, incluyendo copia del Acuerdo:
 - a la Comisión Nacional del Mercado de Valores mediante la comunicación de hecho relevante según lo establecido en el artículo 228 de la Ley de Mercado de Valores y
 - (ii) a GAMESA.
 - 16.1.2 SIEMENS, en virtud de un "ad hoc Mitteilung" según lo establecido en el artículo 15, apartado 1 de la Ley de Mercado de Valores alemana (WpHG), deberá informar sobre los acuerdos alcanzados en relación con la Operación y los principales términos y condiciones del presente Acuerdo, si resultase necesario.
- 16.2 Con carácter inmediato a las comunicaciones a las que se hace referencia en la Cláusula 16.1.1 y adicionalmente de conformidad con el artículo 531 de la Ley de Sociedades de Capital, IBERDROLA depositará una copia de este Acuerdo (junto con una traducción jurada a español) en el Registro Mercantil de Vizcaya.
- Salvo lo dispuesto en relación con los anuncios, comunicaciones y presentaciones en las Cláusulas 16.1 y 16.2, ninguna de las Partes emítirá, sin el consentimiento previo por escrito de la otra Parte, ningún tipo de comunicado de prensa, emitirá documento público o realizará declaraciones públicas o cualquier otro tipo de comunicación que implique la divulgación de información a cualquier persona que no sea parte del presente Acuerdo (a excepción de GAMESA), con anterioridad o posterioridad a la efectividad de la Fusión en relación con los asuntos tratados o relacionados en el presente Acuerdo o cualquier asunto accesorio al mismo. La presente Cláusula 16.3 no será de aplicación a los anuncios o divulgaciones permitidos al amparo de la Cláusula 17 o en virtud del Acuerdo de Fusión.

17 Confidencialidad

- 17.1 Salvo disposición contraria en el presente documento, ninguna de las Partes revelará la información relacionada con el presente Acuerdo. No obstante lo anterior, el presente Acuerdo podrá revelarse:
 - (i) por cualquiera de las Partes intervinientes a sus asesores legales o financieros o empleados o a los de su Grupo según las necesidades y según los términos en que dichos asesores o empleados asuman el compromiso de cumplir con lo dispuesto en la presente Cláusula en relación con dicha información de la misma manera que si formaran parte del mismo;
 - (ii) en el supuesto de que la revelación o utilización de la información se tradujera en un mayor beneficio para las Partes de conformidad con este Acuerdo;
 - se revele o utilice como consecuencia de un auto judicial o laudo arbitral respecto del presente Acuerdo o cualquier otro acuerdo suscrito de conformidad o en virtud del presente Acuerdo;

[página 20]

- (iv) en el supuesto de que el presente Acuerdo devenga de dominio público (y su revelación no se hubiera producido como consecuencia del incumplimiento de las disposiciones contenidas en esta Cláusula);
- (v) en cualquier otra situación por cualquiera de las Partes en el supuesto de que la legislación aplicable o cualquier autoridad gubernamental, judicial o regulatoria así se lo exija (incluido cualquier órgano regulador de mercado de valores o activos) o cualquier otro tipo de órgano con jurisdicción sobre dicha Parte (o sobre cualquier otra sociedad de su Grupo) o normativa en materia de mercado de valores o cualquier sentencia, auto o requisito vinculante que provenga de cualquier autoridad competente; o
- (vi) la otra Parte hubiera otorgado su consentimiento previo por escrito para tal revelación o utilización;

a condición de que, con anterioridad a la revelación o utilización de la información de conformidad con la Cláusula 17(v), en la medida en que fuera legalmente posible, las Partes colaborarán de buena fe para acordar el momento y contenido de dicha revelación o utilización, siempre que sea posible.

18 Socios de SIEMENS Wind Power Parent

- 18.1 IBERDROLA e IBERDROLA PARENT reconocen y aceptan que:
 - 18.1.1 Antes de la efectividad de la Fusión, además de SIEMENS, una o más de las sociedades del Grupo de SIEMENS, podrán también ser socios de SIEMENS Wind Power Parent (cada una de ellas, individualmente, un "Socio de SIEMENS Wind Power Parent" y, conjuntamente, los "Socios de SIEMENS Wind Power Parent") y, consecuentemente, y por virtud de la Fusión, convertirse también en accionistas de GAMESA; y
 - 18.1.2 SIEMENS tendrá derecho, pero no estará obligado, a ceder parcialmente, delegar o transmitir sus derechos y obligaciones bajo este Acuerdo a cualquier Socio de SIEMENS Wind Power Parent antes de la efectividad de la Fusión, sin consentimiento previo de IBERDROLA o de IBERDROLA PARENT.
- En caso de cesión parcial, delegación o transmisión de derechos y obligaciones conforme a la Cláusula 18.1.2 anterior, se entenderá que cualesquiera derechos y obligaciones de SIEMENS bajo este Acuerdo se distribuirán entre SIEMENS y el correspondiente Socio de SIEMENS Wind Power Parent a pro-rata de su participación en el capital social de SIEMENS Wind Power Parent (o, tras la efectividad Fusión, a pro-rata de su participación accionarial en GAMESA), siempre y cuando SIEMENS garantice a IBERDROLA y a IBERDROLA PARENT, solidariamente con los Socios de SIEMENS Wind Power Parent, el cumplimiento y observancia por cualquier Socio de SIEMENS Wind Power Parent de sus obligaciones y compromisos bajo o conforme a este Acuerdo. A efectos aclaratorios, en caso de cesión parcial, delegación o transmisión de derechos y obligaciones conforme a la Cláusula 18.1.2 anterior, cualesquiera referencias a SIEMENS bajo este Acuerdo se entenderán hechas también al/los Socio/s de SIEMENS Wind Power Parent, según el contexto.
- 18.3 Si SIEMENS no lleva a cabo cesión parcial, delegación o transmisión de sus derechos y obligaciones bajo este Acuerdo a favor del correspondiente Socio/s de SIEMENS Wind Power Parent, será de aplicación lo siguiente:

[página 21]

- 18.3.1 dichos derechos y obligaciones permanecerán exclusivamente en SIEMENS;
- 18.3.2 SIEMENS continuará ejercitando cualesquiera derechos de forma individual;
- 18.3.3 SIEMENS podrá, alternativamente, cumplir cualesquiera obligaciones por si misma o si, en la medida requerida, el cumplimiento de cualquier obligación bajo este Acuerdo demandara cualquier acción por parte de cualquier Socio/s de SIEMENS Wind Power Parent, SIEMENS hará que las acciones necesarias sean implementadas por el Socio/s de SIEMENS Wind Power Parent de suerte que las obligaciones bajo este Acuerdo sean cumplidas; y
- 18.3.4 en el entendimiento de que, frente a IBERDROLA e IBERDROLA PARENT, SIEMENS y cualquier Socio/s de SIEMENS Wind Power Parent deben siempre actuar bajo una única voz y de que:
 - (i) Con respecto a cualesquiera obligaciones y responsabilidades asumidas por SIEMENS bajo este Acuerdo, ni IBERDROLA ni IBERDROLA conjuntamente con IBERDROLA PARENT, tendrán acción directa frente a Socio/s de SIEMENS Wind Power Parent sino únicamente frente a SIEMENS; y
 - (ii) Con respecto a cualesquiera derechos de SIEMENS bajo este Acuerdo, se entenderá que IBERDROLA e IBERDROLA PARENT han cumplido íntegramente sus respectivas obligaciones mediante su liquidación frente a SIEMENS.

19 Comité de Seguimiento del Acuerdo de Accionistas

Las Partes acuerdan que, con el fin de realizar un seguimiento de la ejecución de, y cumplimiento con, este Acuerdo, se constituirá un comité específico, que estará compuesto (i) por parte de SIEMENS, por el Consejero Delegado, el Miembro del *Vorstand* de SIEMENS responsable de la división de Energía y el Director de Fusiones y Adquisiciones y (ii) por parte de IBERDROLA, por el Consejero Delegado y el Director de Desarrollo Corporativo de IBERDROLA PARENT.

Mientras este Acuerdo se encuentre en vigor, este comité se reunirá dos veces al año o a solicitud de cualquiera de las Partes. Las reuniones se celebrarán en el lugar que las Partes designen de mutuo acuerdo.

20 Disposiciones adicionales

20.1 Costes

Cada una de las Partes soportará los costes y gastos en los que incurra de conformidad con la preparación, negociación y/o formalización de este Acuerdo.

20.2 Modificaciones y Renuncia

20.2.1 Modificaciones

Las modificaciones realizadas al presente Acuerdo no surtirán efecto a menos que se realicen por escrito y la firmen las Partes o sus debidos representantes.

[página 22]

20.2.2 Ausencia de renuncia

- (i) La falta o retraso por cualquiera de las Partes en el ejercicio de cualquier acción o derecho bajo este Acuerdo o la indulgencia o ampliación del plazo para el ejercicio de dicha acción o derecho, no se entenderá como renuncia al mismo ni el ejercicio simple o parcial de cualquier acción o derecho impedirá el ejercicio de esa u otra acción o derecho.
- (ii) La renuncia por cualquier Parte a un requisito de este Acuerdo o al ejercicio de cualquier acción o derecho bajo este Acuerdo, no surtirá efecto a menos que dicha renuncia se realice por escrito y conste firmada por tal Parte. En el supuesto de renuncia a sus derechos en virtud de un incumplimiento concreto de lo dispuesto en el presente Acuerdo, dicha renuncia no se entenderá como renuncia a los derechos que posteriormente se puedan derivar de la repetición de tal incumplimiento.

20.3 Acuerdo íntegro. Ausencia de actuación concertada

- 20.3.1 El presente Acuerdo (junto con los documentos mencionados en este Acuerdo o suscritos como consecuencia del mismo) representa la totalidad de los acuerdos y compromisos de las Partes y deroga al resto de acuerdos, compromisos o disposiciones anteriores (tanto verbales como por escrito) relacionados con su objeto y el objeto de tales documentos adicionales.
- 20.3.2 Lo dispuesto en el presente Acuerdo no se entenderá como determinante u origen de ningún tipo de actuación concertada entre las Partes con el objetivo de adquirir el control de GAMESA o establecer o imponer ningún tipo de política común en relación con la estrategia o gestión de GAMESA o su Grupo. En concreto, salvo disposición en contrario en el presente Acuerdo y no obstante las obligaciones de IBERDROLA en virtud del mismo:
 - tanto IBERDROLA como SIEMENS serán libres para ejercitar, según estimen conveniente o apropiado, cualquier derecho de voto o de otro tipo derivado de su participación en GAMESA; y
 - (ii) cualquier consejero dominical nombrado por IBERDROLA o SIEMENS dentro del Consejo de Administración tendrá plena libertad para ejercitar su derecho de voto como consejero de GAMESA según estime conveniente y según su criterio en relación con todos los asuntos comerciales.

20.4 Garantías de IBERDROLA PARENT. Ejercicio de derechos

- 20.4.1 IBERDROLA PARENT, de forma incondicional, irrevocable y solidariamente garantiza a SIEMENS el debido y puntual cumplimiento y observancia por parte de IBERDROLA de todas sus obligaciones, compromisos y acuerdos en virtud del presente Acuerdo.
- 20.4.2 Las Partes reconocen que IBERDROLA PARENT es Parte del presente Acuerdo principalmente a los efectos de que la sociedad matriz otorgue las garantías estipuladas en la Cláusula 20.4.1 y asuma determinados compromisos propios previstos en el presente Acuerdo. Por consiguiente, cualquier derecho o reclamación en contra de SIEMENS en virtud del presente Acuerdo se ejercitará por IBERDROLA o por IBERDROLA junto a IBERDROLA PARENT, en el entendido

[página 23]

de que en ningún caso IBERDROLA e IBERDROLA PARENT tendrán derecho a (i) ejercitar cualquier derecho o cualquier reclamación contra SIEMENS por separado o (ii) en caso de ejercitar cualquier derecho o cualquier reclamación contra SIEMENS de forma conjunta, reclamar más de una vez respecto a un mismo asunto.

20.5 Modificación de reglamentos

Una vez la Fusión sea efectiva, SIEMENS e IBERDROLA deberán hacer, en su condición de accionistas de GAMESA y, en la medida en que puedan hacerlo, con respecto a GAMESA, que los Reglamentos del Consejo de Administración, de la Comisión de Nombramientos y Retribuciones y de la Comisión de Auditoría, Cumplimiento y Operaciones Vinculadas, según corresponda y en la medida legalmente posible, sean modificados para modificar en ellos las estipulaciones contenidas en este Acuerdo.

20.6 Cesión

Con excepción de lo dispuesto en la Cláusula 18, ninguna Parte podrá, sin el previo consentimiento por escrito de las otras Partes, ceder cualquiera de sus derechos u obligaciones surgidos en virtud del Acuerdo ni conceder garantía real alguna sobre los mismos o transferir el beneficio de todo o de cualquier parte del Acuerdo.

20.7 Ejemplares

- 20.7.1 Este Acuerdo podrá ser otorgado en cualquier número de ejemplares y por cada Parte en ejemplares separados. Cada ejemplar es un original, pero todos ellos constituirán juntos uno y el mismo instrumento. La entrega de un ejemplar del Acuerdo como adjunto a un correo electrónico o por medio de fax será un modo eficaz de entrega.
- **20.7.2** En el supuesto de que las Partes realicen el Acuerdo en ejemplares de acuerdo con la Cláusula 20.7.1, deberán, tras dicha realización, proporcionar sin demora a la otra Parte los originales firmados.

20.8 Nulidad y circunstancias sobrevenidas

- 20.8.1 Si cualquier disposición de este Acuerdo deviniese ilegal, inválida o no ejecutable en su totalidad o en parte, la disposición se aplicará con cualquier eliminación o modificación que fuese necesaria para que devenga de nuevo legal, válida y ejecutable y de este modo surta efecto a la intención comercial de las Partes.
- 20.8.2 En la medida en que no sea posible eliminar o modificar la disposición, en su totalidad o en parte, de acuerdo con la Cláusula 20.8.1, dicha disposición o parte de ella, en la medida en que sea ilegal, inválida o inexigible, no será considerada parte del Acuerdo, no viéndose afectada la legalidad, validez y exigibilidad del resto del Acuerdo, sin perjuicio de cualquier eliminación o modificación realizada en virtud de la Cláusula 20.8.1.
- 20.8.3 Las Partes recíprocamente declaran y reconocen que sus derechos y obligaciones bajo este Acuerdo y, en particular, bajo la Sección III, también constituyen un factor esencial y decisivo para su ejecución y sus respectivas aportaciones y apoyo a la Operación, tal y como aquí se establece.

En este sentido, en caso de que sobreviniera cualquier Requisito (según este término se define más abajo) vinculante de carácter regulatorio que entrase en

[página 24]

conflicto con cualesquiera derechos y/u obligaciones de cualquiera de las Parteś bajo este Acuerdo (un "Conflicto"), las Partes se comprometen a negociar de buena fe las mínimas modificaciones del presente Acuerdo, con el fin de hacerlo compatible con el correspondiente Requisito vinculante de carácter regulatorio, preservando en todo momento el mismo equilibrio y espíritu subyacentes (las "Negociaciones de Buena Fe").

A estos efectos, SIEMENS manifiesta y garantiza a IBERDROLA que (i) una de las condiciones suspensivas a las que está sujeto el Acuerdo de Fusión consiste en el otorgamiento por la CNMV de la exención referida en el artículo 8.g) del Real Decreto 1066/2007, de 27 de julio, sobre el régimen de las ofertas públicas de adquisición de valores, y que (ii) para los casos en que la CNMV indique que dicha exención únicamente puede ser otorgada a SIEMENS si (a) ciertas restricciones u obligaciones son impuestos o (b) ciertos compromisos son ofrecidos (tanto (a) como (b), el "Requisito"), de acuerdo con los términos del Acuerdo de Fusión, SIEMENS no estará obligada a aceptar ninguna restricción o pacto o a ofrecer ningún compromiso que se considere indebidamente gravoso.

- (i) En caso de que el Conflicto fuese relativo a (1) cualquiera de los derechos y/u obligaciones de las Partes en virtud de la Sección III o (2) las Cláusulas 5 o 20.8 de este Acuerdo, las siguientes estipulaciones serán de aplicación:
 - (a) Las Negociaciones de Buena Fe se llevarán a cabo dentro de un plazo de quince (15) Días Hábiles a contar desde la fecha en la que cualquiera de las Partes curse primera notificación a la otra Parte con el fin de iniciar las negociaciones (el "Primer Periodo de Negociación").
 - (b) En caso de que, dentro del Primer Periodo de Negociación las Partes alcancen un acuerdo relativo a las modificaciones a este Acuerdo necesarias para hacerlo compatible con el Requisito, deberán implementar dichas modificaciones con carácter inmediato y SIEMENS deberá aceptar u ofrecer, según corresponda, el Requisito.
 - (c) Si las Partes no alcanzan un acuerdo dentro del Primer Periodo de Negociación, será de aplicación lo siguiente:
 - (I) Dentro de los cinco (5) Días Hábiles siguientes a la finalización del Primer Periodo de Negociación (el "Primer Periodo de Notificación de Terminación"), SIEMENS deberá notificar a IBERDROLA si pretende, o no, aceptar u ofrecer, según corresponda, el Requisito. Una vez esta notificación haya sido cursada, los Consejeros Delegados tanto de IBERDROLA PARENT como de SIEMENS deberán reunirse con el fin de tratar de llegar a un acuerdo dentro de un plazo de diez (10) Días Hábiles a contar desde la fecha en la que dicha notificación fuera cursada por SIEMENS a IBERDROLA (el "Segundo Periodo de Negociación").

Si no se pudiera alcanzar un acuerdo dentro del Segundo Periodo de Negociación y SIEMENS decidiera, de forma [página 25]

irrevocable y definitiva, que (i) no aceptará u ofrecerá, según corresponda, el Requisito, y que, adicionalmente, (ii) no renunciará a la condición suspensiva a la que está sujeta la Fusión consistente la exención referida en el artículo 8.g) del Real Decreto 1066/2007, de 27 de julio, sobre el régimen de las ofertas públicas de adquisición de valores (la "Decisión de Terminación"), SIEMENS deberá notificar a IBERDROLA y a GAMESA la Decisión de Terminación sin dilación injustificada.

La Decisión de Terminación será definitiva e irrevocable y SIEMENS se compromete a, y manifiesta y garantiza a IBERDROLA, que la terminación tanto del Acuerdo de Fusión como de la Fusión será efectiva, en cualquier caso, (a) en la fecha en que GAMESA acepte dicha terminación, si dicha fecha es anterior a la fecha límite prevista en el Acuerdo de Fusión para el cumplimiento de la condición suspensiva arriba referida, o (b) en dicha fecha límite, si GAMESA no acepta la terminación antes de dicha fecha (tanto (a) como (b), la "Fecha de Terminación de la Fusión").

- (II)En caso de que, dentro de los cinco (5) Días Hábiles siguientes a la finalización del Segundo Periodo de Negociación (el "Periodo de Ejercicio"), SIEMENS no haya notificado la Decisión de Terminación a IBERDROLA o, por cualquier razón, el Acuerdo de Fusión y la Fusión no hayan sido terminados, IBERDROLA tendrá el derecho de solicitar a SIEMENS que, dentro de los cinco (5) Días Hábiles siguientes a la finalización del Periodo de Ejercicio (el "Segundo Periodo de Notificación de Terminación"), SIEMENS notifique formalmente a GAMESA su Decisión de Terminación y ofrezca a GAMESA terminar con carácter inmediato el Acuerdo de Fusión y la Fusión, en cuyo caso, IBERDROLA tendrá derecho a resolver este Acuerdo En estos casos, la terminación tanto del Acuerdo de Fusión como de la Fusión será efectiva en la Fecha de Terminación de la Fusión.
- (III) Si SIEMENS no cumple con la solicitud efectuada por IBERDROLA conforme al párrafo (II) anterior (incluyendo, a efectos aclaratorios, el supuesto de que SIEMENS no termine el Acuerdo de Fusión y la Fusión si así lo acepta GAMESA tras la recepción de la Decisión de Terminación de SIEMENS) dentro de los cinco (5) Días Hábiles siguientes a la finalización del Segundo Periodo de Notificación de Terminación, esto se considerará un Incumplimiento esencial de SIEMENS a los efectos de lo establecido en la Cláusula 12 e IBERDROLA estará facultada para ejercitar sus derechos bajo dicha Cláusula 12. Dichos derechos serán exigibles desde la Fecha de Terminación de la Fusión.

[página 26]

Este párrafo (III) será también de aplicación, y los derechos otorgados bajo la Cláusula 12 también serán ejercitables y exigibles por IBERDROLA desde la Fecha de Terminación de la Fusión, si, en dicha fecha, SIEMENS no ha resuelto el Acuerdo de Fusión y la Fusión y por tanto, el Acuerdo de Fusión y el proceso de Fusión continúan vigentes tras la Fecha de Terminación de la Fusión, lo que se considerará un Incumplimiento esencial.

- (ii) En caso de que el Conflicto fuese relativo a cualesquiera derechos y/u obligaciones de las Partes bajo cualquier estipulación de este Acuerdo distinta de las contenidas en la Cláusula 20.8.3(i), será de aplicación lo siguiente:
 - (a) Las Partes se comprometen a llevar a cabo Negociaciones de Buena Fe con el fin de eliminar el Conflicto, haciendo para ello la estipulación o estipulaciones conflictivas compatible(s) con el correspondiente Requisito, siempre preservando el mismo equilibrio y espíritu subyacentes.
 - (b) La efectividad de la estipulación o estipulaciones correspondiente(s) de este Acuerdo que cause el Conflicto quedará suspendida desde el momento en que surja el Conflicto y, en consecuencia, no habrá incumplimiento por ninguna de las Partes en tanto en cuanto, y hasta que, las Partes hayan llegado a un acuerdo de conformidad con lo establecido en la Cláusula 20.8.3(ii)(a) anterior. Como consecuencia de dicha suspensión, el conflicto correspondiente entre este Acuerdo y el Requisito correspondiente se entenderá solventado y no impedirá a SIEMENS completar la Fusión de conformidad con el Acuerdo de Fusión.
- (iii) En el caso de que la duración de los distintos plazos referidos en los párrafos (i) y (ii) anteriores sea incompatible con cualesquiera plazos legales y/o el plazo concedido a la CNMV para resolver sobre la exención referida en el artículo 8.g) del Real Decreto 1066/2007, de 27 de julio, sobre el régimen de las ofertas públicas de adquisición de valores en relación con la Fusión, las Partes acuerdan adaptar dichos plazos en consonancia.
- (iv) Para evitar cualquier duda, las obligaciones de SIEMENS bajo esta Cláusula 20.8.3 existirán sin perjuicio del derecho de SIEMENS a renunciar a la condición suspensiva a la que está sujeta la Fusión, consistente en la exención referida en el artículo 8.g) del Real Decreto 1066/2007, de 27 de julio, sobre el régimen de las ofertas públicas de adquisición de valores. Las Partes acuerdan que, en estos casos, si SIEMENS formulase una oferta con respecto a las acciones de GAMESA, IBERDROLA no estará sujeta a limitación alguna y será libre para evaluar y decidir si apoya, o no, dicha oferta, sin ninguna restricción de cualquier clase.

[página 27]

20.9 Comunicaciones

- Cualquier notificación o comunicación relativa al presente Acuerdo (cada una, la 20.9.1 "Comunicación") deberá realizarse:
 - (i) por escrito y en inglés; y
 - (ii) entregarse en mano, por fax, correo certificado, correo electrónico, o a través de una empresa de mensajería con reconocimiento internacional.
- La Comunicación a cualquier otra Parte se le deberá enviar a la siguiente dirección, o aquella persona o dirección que la correspondiente Parte notifique a las otras Partes:
 - Si el envío se realiza a SIEMENS: (i)

Destinatario:

D. Roland Meinzer (Senior Vice President Post-Closing

Management)

St.-Martin-Str. 76

81541 Múnich

Alemania

cc: D. Anton Steiger (Chief Counsel Corporate and M&A)

Werner-von-Siemens Strasse 50

91052 Erlangen

Alemania

Correo

roland.meinzer@siemens.com

electrónico:

cc: anton.steiger@siemens.com

(ii) Si el envío se realiza a IBERDROLA y/o a IBERDROLA PARENT

Destinatario: D. Manuel Toledano

calle Tomás Redondo 1

28033 Madrid

España

Correo

mtoledano@iberdrola.es

electrónico:

- 20.9.3 Cualquiera de las Partes podrá cambiar la dirección facilitada para Comunicaciones siempre que se le notifique a las otras Partes con un plazo de preaviso no inferior a cinco (5) días de acuerdo con lo dispuesto en esta cláusula. Hasta la finalización de dicho periodo, las Comunicaciones realizadas a dichas direcciones serán válidas.
- 20.10 Sometimiento a fuero y legislación aplicable
 - 20.10.1 Fuero aplicable

Este Acuerdo y cualquier obligación no contractual surgida de o en conexión con el mismo, se regirá e interpretará según lo dispuesto por la ley común española.

20.10.2 Legislación aplicable

MARÍA DÁVILA BLANCO
Traductora - Intérprete Jurada de INGLÉS
Nº 6181

[página 28]

Las Partes acuerdan de forma irrevocable someter a la jurisdicción y competencia de los Juzgados y Tribunales de Madrid capital, cualquier conflicto que pudiera surgir de o en relación con el Acuerdo y los documentos que deberán suscribirse conforme a lo previsto en él y que, en consecuencia, cualquier procedimiento que surja de o en relación con el Acuerdo se llevará ante dichos Tribunales. Las Partes se someten de forma irrevocable a dichos tribunales y renuncian a formular cualquier objeción a que los procedimientos se sigan ante dichos tribunales por razón del fuero que pudiera corresponderles o sobre la base de que el procedimiento se ha iniciado en fuero equivocado.

MARÍA DÁVILA BLANCO
Traductora - Intérprete Jurada de INOLES
Nº 6181

[página 29]

En prueba de conformidad con cuanto antecede, las Partes firman este Acuerdo, en la fecha indicada en la primera página.

[consta firma ilegible]
FIRMADO por Dña. Dagmar Mundani
En su condición de apoderado (<i>Prokuristin</i>) en nombre y representación de SIEMENS AKTIENGESELLSCHAFT
[consta firma ilegible]
FIRMADO por Dr. Christian Zentner
En su condición de apoderado (<i>Prokurist</i>) en nombre y representación de SIEMENS AKTIENGESELLSCHAFT
[consta firma ilegible]
FIRMADO por D. Javier Salazar Blanco
En su condición de administrador mancomunado en nombre y representación de IBERDROLA PARTICIPACIONES, S.A.U.
[consta firma ilegible]
FIRMADO-por-D. Gerardo-Codes-Calatrava
En su condición de administrador mancomunado en nombre y representación de IBERDROLA PARTICIPACIONES, S.A.U.
[consta firma ilegible]
FIRMADO por D. Pedro Javier Azagra Blázquez
En su condición de apoderado mancomunado en nombre y representación de IBERDROLA, S.A.

MARÍA DÁVILA BLANCO
Traductora - Intérprete Jurada de INGLÉS
Nº 6181

[página	30]

[consta firma ilegible]

FIRMADO por D. Juan Carlos Rebollo Liceaga

En su condición de apoderado mancomunado en nombre y representación de IBERDROLA, S.A.

[página 31]

Anexo 1.1 Definiciones

Las siguientes palabras y expresiones han sido utilizadas en el Acuerdo teniendo el significado fijado a continuación:

- "Acuerdo" tendrá el significado dado en el Considerando (E);
- "Consejo de Administración" tendrá el significado dado en la Cláusula 8.1.1;
- "Incumplimiento" tendrá el significado dado en la Cláusula 12.1;
- "Notificación de Incumplimiento" tendrá el significado dado en la Cláusula 12.3.1;
- "**Día Hábil**" significa cualquier día de la semana exceptuando sábado, domingo o festividades en España o en Alemania;
- "Conflicto" tendrá el significado dado en la Cláusula 20.8.3;
- "CNMV" significa Comisión Nacional del Mercado de Valores;
- "Proyecto Común de Fusión" tendrá el significado dado en la Cláusula 3.1.4;
- "Plazo de Subsanación" tendrá el significado dado en la Cláusula 12.3.2
- "Periodo de Ejercicio" tendrá el significado dado en la Cláusula 20.8.3.
- "Primer Periodo de Negociación" tendrá el significado dado en la Cláusula 20.8,3.
- "Primer Periodo de Notificación de Terminación" tendrá el significado dado en la Cláusula 20.8.3.
- "Precio de Adquisición Final" tendrá el significado dado en la Cláusula 12.5.
- "Precio Mínimo" tendrá el significado dado en el párrafo (i) de la Cláusula 11.1;
- "GAMESA" tendrá el significado dado en el Considerando (A);
- "Negociaciones de Buena Fe" tendrá el significado dado en la Cláusula 20.8.3;
- "**Grupo**" significa con respecto a una sociedad, dicha sociedad y cualquier empresa que sea filial de la sociedad en cada momento:
- "IBERDROLA" tendrá el significado dado en la sección (2) de las Partes;
- "IBERDROLA PARENT" tendrá el significado dado en la sección (3) de las Partes;
- "Periodo de Integración" tendrá el significado dado en la Cláusula 8.5;
- "Comité de Seguimiento de la Integración" tendrá el significado dado en la Cláusula 8.8.1;
- "Periodo Interino" tendrá el significado dado en la Cláusula 4;
- "Fusión" tendrá el significado dado en el Considerando (D);
- "Acuerdo de Fusión" tendrá el significado dado en el Considerando (D);
- "Fecha de Terminación de la Fusión" tendrá el significado dado en la Cláusula 20.8.3;
- "Comunicación" tendrá el significado establecido en la Cláusula 20.9.1;
- "Oferta" tendrá el significado dado en la Cláusula 11.2.2(i)
- "Fecha de Cierre de la Oferta" tendrá el significado dado en la Cláusula 11.2.2;

[página 32]

- "Comunicación de la Oferta" tendrá el significado dado en la Cláusula 11.2.1;
- "Partes" hará referencia a SIEMENS, IBERDROLA e IBERDROLA PARENT en conjunto como las "Partes" y a cada una de ellas de forma individual como una "Parte".
- "Precio de Adquisición Provisional" tendrá el significado dado en la Cláusula 12.5.
- "Actividad Prohibida" tendrá el significado dado en la Cláusula 6.1.2;
- "Comunicación de Compra" tendrá el significado establecido en la Cláusula 11.2.2;
- "Condición" tendrá el significado establecido en la Cláusula 20.8.3;
- "Segundo Periodo de Negociación" tendrá el significado dado en la Cláusula 20.8.3.
- "Segundo Periodo de Notificación de Terminación" tendrá el significado dado en la Cláusula 20.8.3.
- "SIEMENS" tendrá el significado dado en la sección (1) de las Partes;
- "Negocio Eólico de SIEMENS" tendrá el significado dado en el Considerando (B);
- "Matriz de Energía Eólica de SIEMENS" tendrá el significado dado en la Cláusula 3.1.2;
- "Socio de SIEMENS Wind Power Parent" tendrá el significado dado en la Cláusula 18.1.1;
- "Régimen de las Ofertas Públicas de Adquisición de Valores" se refiere al Real Decreto 1066/2007, de 27 de julio, sobre el régimen de las ofertas públicas de adquisición de valores, vigente en cada momento:
- "Decisión de Terminación" tendrá el significado dado en la Cláusula 20.8.3.
- "Tercero" tendrá el significado dado en la Cláusula 11.2.6;
- "Operación de Terceros" significa una propuesta respecto a una oferta, fusión o combinación de negocios u operaciones similares (sujetas o no a cualquier condición previa), propuesta por un tercero y cuyo objetivo es habilitar a ese tercero (o a cualquier otra persona) a adquirir la totalidad o una parte significativa de las acciones de GAMESA o la totalidad o una parte significativa de los activos o negocio de GAMESA, o cualquier otro acuerdo u operación o una serie de ellos que pueda ser inconsistente con, o perjudicial para, la implementación de la Fusión:
- "Activos Totales" tendrá-el-significado-dado-en-la-Cláusula-10(ii)-
- "Operación" tendrá el significado dado en el Considerando (D);
- "Transmisión" tendrá el significado establecido en la Cláusula 11.2;
- "Fecha de Transmisión" tendrá el significado dado en la Cláusula 12.4.
- "Acciones sujetas a Transmisión" tendrá el significado establecido en la Cláusula 11.2;
- "Valorador" deberá ser un reputado banco de inversión internacional, seleccionado conjuntamente por las Partes en el plazo de cinco (5) Días Hábiles desde la terminación del Plazo de Subsanación. Si las Partes no llegaran a un acuerdo en dicho plazo, SIEMENS escogerá uno de entre una lista de tres (3) reputados bancos de inversión internacional con voluntad de aceptar el mandato propuestos por IBERDROLA y si SIEMENS rechaza hacer esa elección, IBERDROLA tendrá derecho a hacerla.

[constan dos visados en todas las páginas del documento salvo en el bloque de firmas]

Doña María Dávila Blanco, Traductora-Intérprete Jurada de inglés nombrada por el Ministerio de Asuntos Exteriores y de Cooperación, certifica que la que antecede es traducción fiel y completa al español de un documento redactado en inglés.

En Madrid, a 17 de junio de 2016

Fdo.: María Dávila Blanco